

RESEARCH ARTICLE

# A Methodological Framework for the Integrated Design of Decision-Intensive Care Pathways—an Application to the Management of COPD Patients

Carlo Combi<sup>1</sup> · Barbara Oliboni<sup>1</sup> · Alessandro Zardini<sup>2</sup> · Francesca Zerbato<sup>1</sup>

Received: 23 January 2017 / Revised: 21 September 2017 / Accepted: 2 October 2017 / Published online: 17 October 2017 © Springer International Publishing AG 2017

Abstract Healthcare processes are by nature complex, mostly due to their multidisciplinary character that requires continuous coordination between care providers. They encompass both organizational and clinical tasks, the latter ones driven by medical knowledge, which is inherently incomplete and distributed among people having different expertise and roles. Care pathways refer to planning and coordination of care processes related to specific groups of patients in a given setting. The goal in defining and following care pathways is to improve the quality of care in terms of patient satisfaction, costs reduction, and medical outcome. Thus, care pathways are a promising methodological tool for standardizing care and decision-making. Business process management techniques can successfully be used for representing organizational aspects of care pathways in a standard, readable, and accessible way, while supporting process development, analysis, and re-engineering. In this paper, we introduce a methodological framework that fosters the integrated design, implementation, and enactment of care processes and related decisions, while considering proper representation and management of organizational and clinical information. We focus here and discuss in detail the design phase, which encompasses the simulation of care pathways. We show how business process model and notation (BPMN) and decision model and notation (DMN) can be combined for supporting intertwined aspects of decision-intensive care pathways. As a proof-of-concept, the proposed methodology

Francesca Zerbato francesca.zerbato@univr.it

<sup>&</sup>lt;sup>1</sup> Department of Computer Science, University of Verona, Strada Le Grazie, 15, 37134 Verona, Italy

<sup>&</sup>lt;sup>2</sup> Department of Business Economics, University of Verona, Via Cantarane, 24, 37129 Verona, Italy

has been applied to design care pathways related to chronic obstructive pulmonary disease (COPD) in the region of Veneto, in Italy.

**Keywords** Care pathways · Decision modeling · Methodological framework · Business process model and notation · Decision model and notation · Healthcare process modeling

# 1 Introduction

Healthcare systems are among the most complex and challenging organizational environments [1]. Healthcare organizations are becoming more and more fragmented into specialized and integrated functional units, which cooperate and exchange information to provide primary, secondary, and specialist care [2]. Healthcare organizations operate by enacting a wide range of processes with different characteristics and requirements, spanning from those capturing clinical procedures, such as diagnosis and treatment, to those representing organizational and administrative tasks, such as examination planning and patient transfer [3]. Yet, clinical and organizational processes are not independent, but rather complementary and intertwined.

Among the many kinds of healthcare processes, *care pathways* received increasing attention in the last years, as they consider both organizational requirements and clinical tasks. Care pathways have been defined in slightly different ways, according to different perspectives. Here we consider two widely acknowledged definitions. The first one specifies that "A care pathway is a complex intervention for the mutual decision-making and organization of care processes for a well-defined group of patients during a well-defined period" [4], while in [5] integrated care pathways are defined as "structured multidisciplinary care plans which detail essential steps in the care of patients with a specific clinical problem." Both definitions contain some common elements that characterize care pathways. Among them, we underline here the relevance related to the definition of care processes/plans, to the multidisciplinarity of such plans, the focus on specific groups of patients, and the decision-making aspects related to such processes.

Thus, the computer-based design and management of care pathways needs to be supported by methodologies and technologies, where all the considered aspects are suitably addressed. In this context, the need to design and enact standard and reliable healthcare processes, able to reduce the occurrence of exceptional events that would compromise the quality and the efficiency of organizational and clinical outcomes, is not new to the domain of healthcare process management [1, 3, 6].

Among existing technologies, business process systems emerged as a leading technology also in the healthcare domain [3, 7-11].

According to recent reports, the global business process management (BPM) market is accounted for \$5.51 billion in 2015 and is expected to reach \$ 17.96 billion by 2022 [12], and \$ 23.04 billion by 2024 [13].

Healthcare is among the main application domains (verticals, in the report terminology) of BPM technology, while process modeling (per se) is one of the main focuses of BPM tools [13]. Moreover, patient monitoring in hospitals is underlined as one of the leading applications of BPM solutions in the healthcare industry. The widely increasing healthcare-based regulations are indicated as one of the leverages to increase the business process management market growth in the healthcare industry [13]. To complete the picture, among the key players in this market we find Oracle, Microsoft, IBM, SAP, and many others [12].

Among the plethora of existing business process languages [14], the business process model and notation (BPMN 2.0) is currently the leading standard for process design, developed by the object management group (OMG) [15]. BPMN provides a consolidated graphical notation used to outline process activities, their logical and temporal ordering, and the resources responsible for their execution. BPMN diagrams are suitable to design, visualize, and document processes at different levels of abstraction and according to different perspectives (process structure, interaction between processes, and so on), thus meeting the needs of various categories of users and fostering both process engines [16–18] that allow stakeholders to actively monitor executable process at run-time and support the interaction with knowledge workers through domain-specific client and invoked applications.

Indeed, BPM systems provide rich development environments that integrate design with process simulation, which encompasses techniques used to walk through a process in a step-by-step manner in order to check whether the process actually behaves as desired [19]. In BPM, simulation is the most commonly used approach for quantitative process analysis [20], as it enables the identification of shortcomings and procedural bottlenecks, thus allowing process managers to assess quality, check conformance, and to progressively tune organizational outcomes.

Recently, the OMG proposed the decision model and notation (DMN 1.1) standard [21] to complement BPMN for supporting decision design, management and implementation, according to the well-known separation of concerns principle [22]. DMN diagrams are meant to model decisions by representing input information, the used knowledge models, and their requirements for decision logic. Such explanatory diagrams portray the structure of one or more decisions and can be linked to one or more processes or activities, depending on the scope of decision-making. Thus, coupling BPMN processes and DMN diagrams could be beneficial to support the integrated design of decision-intensive care pathways, where complex clinical and organizational knowledge for decision-making needs to be represented.

Despite the presence of such advanced models and software tools, at the best of our knowledge, only few attempts have been made to manage care pathways through the integrated use of BPMN and DMN and, even more important, no attention has been paid to propose/extend development methodologies for the specific context of care pathways. Thus, according to the depicted scenario, in this paper we propose a new methodological framework to support the integrated design, implementation, and enactment of decision-intensive care pathways under an organization standpoint in order to foster the modeling and re-engineering of care pathways and related decision-making through proper information management and data re-use. The methodology defines both the main phases of process development, which are based on BPM best practices [19], and the widely known and adopted modeling languages and tools to be used. In detail, we show how the DMN notation can be used to complement BPMN with the goal of increasing support toward decision-making, hence improving its suitability for healthcare process modeling and enactment. For each phase, relevant information sources are also highlighted. Within the proposed methodology, we will mainly focus on the design phase.

The main novelties, specific for the management of care pathways, of the proposed methodological approach are:

- Supporting the integrated design of care pathways and decisions, based on conjunct use of the standard BPMN and DMN languages;
- Strongly integrating care pathway simulation within the design phase, to support agile development;
- Proposing a classification of the main kinds of information involved in different phases of process and decision development, and investigating the role of databased indicators;
- Using and possibly extending BPMN modeling constructs to represent complex temporal behaviors/ constraints in care pathways;
- Illustrating meaningful aspects of clinical decision-making and evaluating their scope and impact on the overall care pathway.

Moreover, as a further added value, we show a real-world application of the proposed approach to the design of care pathways for chronic obstructive pulmonary disease (COPD), within a regional context.

The presented methodological framework extends the approach introduced in [10] and tackles several new original issues. This paper illustrates a more general methodology that encompasses all the phases of a typical business process life-cycle, thus considering also process implementation and enactment. Moreover, the role of simulation within the design phase is explicitly discussed. In addition, connection points with existing formalisms are highlighted, as the methodology aims to provide a main standard core that can be extended and integrated with context-specific decision support systems and tools, suitable for dealing with complex clinical aspects. Information management is now included in the methodological approach and a classification of data that are relevant for process execution is presented. As for the application to the clinical domain of COPD, the goal of this paper is to illustrate process and decision design for care pathway development, thus covering all the aspects related to requirements analysis, modeling, and simulation. In detail, primary, specialist, and secondary care of COPD have been studied and modeled exhaustively, with the help of stakeholders. Specifically, the presented BPMN and DMN diagrams describe not only diagnosis and assessment of COPD patients, but also ordinary and extraordinary care. Finally, real data have been obtained from currently enacted healthcare processes and they have been used for process simulation in order to analyze as-is organizational procedures from a re-engineering perspective.

The remainder of the paper is organized as follows. Section 2 introduces the fundamentals of the BPMN and DMN standards, elaborating on their integration. Section 3 discusses the proposed methodological framework. Section 4 explains important aspects of COPD treatment and management and illustrates how the introduced methodology is applied to the design and management of care pathways for COPD in the region of Veneto, in northern Italy. Section 5 discusses related work, considering proposals belonging to the domains of BPM and clinical guidelines development. Finally, Section 6 sketches some conclusions and presents future work.

#### 2 Background: Modeling Processes and Related Decisions

In this section, the business process model and notation [15] and the decision model and notation [21] standards are introduced, in turn. Then, the integration of decision and process management is discussed, with particular focus on the new insights provided by the combined use of BPMN and DMN.

#### 2.1 Business Process Model and Notation

In general, a business process (BP) is defined as "a set of activities that are performed in coordination in an organizational and technical environment" [19].

The business process model and notation (BPMN 2.0) was developed under the OMG umbrella as a standard language for graphically representing business processes [15]. Born as a general purpose language, the notation is suitable for modeling processes in multiple application domains and at different levels of abstraction, thanks to its extensible meta-model organized in layers. BPMN provides a set of basic elements meant to be understood by both process designers and practitioners, without extensive training. Yet, the notation includes more advanced constructs and multiple kinds of diagrams that allow expert users to express complex concepts, such as message-based communication, event-processing, and exception handling.

In BPMN, a process is represented by means of a graphical diagram, which is mainly composed of activities, sub-processes, and events. These are connected by a sequence flow that denotes the causal relations between them and is routed by gateways [15]. The basic elements of a process diagram are *activities, events*, and *gateways*. Besides, *swimlanes* are used to represent organizational aspects and process participants, whereas *data artifacts* outline process-related information.

- Activities represent work that is carried out during process execution. BPMN provides two main elements for describing activities, namely, tasks and sub-processes. Tasks are atomic units of work that cannot be broken down to a finer level of abstraction. Graphically, they are depicted as rectangles with rounded corners, enclosing the name of the activity. A marker can be added to detail the type of task, which is associated to a well-defined behavior (e.g., "user tasks" denote that work is assigned to a human performer). Conversely, sub-processes represent compound activities, whose internal detail may (expanded sub-process) or may not (collapsed sub-process) be shown. Collapsed sub-processes are represented as tasks decorated with a "+" maker. Expanded sub-processes consist of an enlarged border that contains all the sub-process elements.
- Events represent facts that occur instantaneously during process execution.
   Graphically, they are depicted as circles and may contain a marker that diversifies the kind of event trigger (e.g., message, timer, signal, etc.). Depending on their behavior, events can either throw a trigger or catch a result. BPMN defines three

kinds of events, based on their position in the process. Start events (single border) initiate a process instance, end events (single thick border) conclude it, and intermediate events (double border) are located at any point between the process start and end. Intermediate events can be attached to the boundary of an activity and they can either be interrupting or non-interrupting. Interrupting boundary events interrupt the activity they are attached to, whereas non-interrupting boundary events initiate a parallel process path.

- Gateways control the divergence and convergence of the sequence flow, according to data-based conditions or event occurrence. Graphically, they are depicted as diamonds with an internal marker that differentiates their routing behavior. The "+" marker is used for parallel branching, while a "×" identifies an exclusive gateway, i.e., a point in the process where a condition must be evaluated in order to choose one path out of more.
- Swimlanes are containers identifying organizational aspects, such as the organization structure (departments, roles) and resources (process participants). Pools are used to represent resource classes or roles, such as, for instance, "Pharmaceutical supplier". In BPMN, roles are used to classify process participants according to their skills, knowledge, and responsibilities within the process. Lanes are used to sub-partition an organization into functional units or individual process participants, and can be arbitrarily nested to denote role hierarchy.
- Data artifacts represent information that is produced or consumed during process execution. In particular, volatile data are represented through *data objects*, which can be exchanged between activities. Graphically, they are depicted as paper documents, with a bent corner. Persistent data sources are modeled as *data stores*, which are associated to activities and can be referenced from different parts of the same process.

Other specific BPMN elements used in this paper will be explained later, when needed.

To introduce the BPMN notation, we employ a simple example taken from the clinical domain of COPD [23]. In particular, let us consider the 6-min walk test (6MWT), one of the clinical tests used to objectively evaluate the functional exercise capacity in COPD patients [24]. 6MWT is a practical test that "measures the distance that a patient can quickly walk on a flat, hard surface in a period of 6 minutes" [24]. The goal for the patient is to walk as far as possible in the given time frame, assisted by a trained technician. This test is typically used to evaluate exercise tolerance in chronic respiratory disease and heart failure, and it allows physicians to monitor pulmonary and cardiovascular functions, among others. 6MWT is typically performed in a dedicated hospital hallway, as timely and appropriate emergency measures must be enacted in case of complications. 6MWT must be immediately stopped if the patient shows signs of chest pain, intolerable dyspnea, leg cramps, staggering, diaphoresis, and pale or ashen appearance [24]. In this scenario, the trained technician must take blood pressure, pulse rate, and oxygen saturation. Oxygen should be given when needed.

A simplified version of the main steps of 6MWT is depicted in the BPMN process diagram of Fig. 1.



Fig. 1 BPMN diagram summarizing some preliminary stages for the 6-min walk test, simplified and adapted from [24]

The goal of Fig. 1 is to exemplify how BPMN can be used for providing a standard representation of clinical procedures and of resources allocated to activities.

After being approved by a physician, the test is conducted in a respiratory diseases unit, represented as a BPMN *pool* and it is carried out by a trained technician, depicted as a *lane*. First of all, the technician should Allow the patient to sit on a chair to rest for at least 10 min before the test starts. This temporal span is represented by an intermediate timer event. A couple of parallel gateways is employed to denote that, during this time frame, the technician must check patient clothing and evaluate relative contraindications. The evaluation of contraindications is a decision about the conduct of the test, based on the current values of the patient vital parameters. The output of the decision is used by the following *exclusive gateway* Can the test be performed? If the test is not believed to put the patient at risk, pulse oximetry can optionally be performed. Exclusive gateway Perform pulse oximetry? is used to represent this option. If it is performed, the physician must measure oxygen saturation; otherwise, this *task* is skipped. If measured, the value of oxygen saturation is recorded in the patient medical record, portrayed as a *data store*. Finally, 6MWT is performed. This is represented by means of a *collapsed process*, as individual steps are not detailed.

However, if the patient experiences dyspnea, chest pain, leg cramps, staggering, diaphoresis, or pale appearance, the 6MWT must be immediately interrupted [24]. In Fig. 1, this is captured by boundary interrupting conditional event stop test, which encodes the aforementioned conditions and generates an exception flow that leads to task assist patient.

Then, the depicted process is concluded by a simple end event.

In general, BPMN can be used in various organization domain for documentation, analysis, and standardization purposes. In healthcare working environments, resources are often having complementary expertise and are loaded with heterogeneous tasks and responsibilities. In this setting, process modeling techniques are useful to improve coordination among different functional units, support communication between participants, and provide overall activity standardization, especially in terms of organization aspects.

BPMN is supported by several (open-source) process modeling tools and customizable engines [16–18] and it is widely used in the context of business management. The main advantage of using BPMN for healthcare process modeling is the fact that it is a well-maintained consolidated standard, complemented by other standard notations for decision modeling [21], case management [25], and process exchange [26]. Moreover, as healthcare processes are often monitored by administrative staff and information technology (IT) experts, it is likely that these users are already familiar with the notation.

Being a graph-oriented process modeling language, BPMN allows combining flow objects almost arbitrarily, thus leading to the definition of hardly readable and complex process diagrams [27]. Large and articulated models are difficult to be used and understood by less trained users and they have proven to be more error-prone. Nevertheless, structural restrictions and modeling guidelines can be applied to increase readability and prevent undesired execution deadlocks [28]. A BPMN process is said to be *well-structured* if for every split node, i.e., a node with multiple outgoing edges, there is a corresponding merge node, i.e., a node with multiple incoming edges, such that the set of nodes delimited by the split and the join nodes form a single-entry-single-exit (SESE) region, and these regions within the process are properly nested [29]. The previously discussed process of Fig. 1 is an example of well-structured diagram. Structured processes have also proven to enhance modularity, which is the ability to decompose the process into smaller reusable components. These often correspond to process sub-regions having specific properties defined over them, which is desirable under a re-engineering standpoint.

In addition, comprehensible diagrams can be used for documentation purposes and as a starting point for communicating with clinicians, during interviews or reengineering interventions.

#### 2.2 Decision Model and Notation

In business process design, decision-making covers a crucial role. Process gateways and conditional events often encode data-based operational decisions, whereas decision activities encompass more complex non-operative decision-making. Decisions affect the process flow and, consequently, they impact both the performance of the process and the quality of its (business) outcomes. However, important decisions often remain hidden within process diagrams and BPMN is misused for decision modeling, despite data-based preconditions of BPMN elements are hard to understand [30, 31]. For these reasons, business process modeling must be properly complemented with decision design.

To respond to this need, the OMG recently developed the standard decision model and notation (DMN 1.1) [21]. DMN has been proposed as a standard notation to bridge the gap between the high level business logic, represented in processes, and decision implementation, typically specified in terms of decision logic. The notation supports the design of both human and automated decision-making, capturing both decision requirements and decision logic. Together, these two levels provide a complete decision model, which complements a process diagram by detailing the decision-making carried out in process tasks [21].

The decision requirements level is modeled by means of a decision requirement graph (DRG), which may be linked to one or more tasks belonging to one or multiple business processes. A DRG represents the main elements involved in a decisionmaking domain and the dependencies between them. As DRGs may be large and complex, they can split into one or more decision requirements diagrams (DRDs). DRDs may be used to show partial views of the whole decision domain, focusing on areas of interest for a specific user or goal. These diagrams define which decisions are made in process tasks, which are their interrelationships, and their requirements for decision logic.

The main elements forming a DRD are introduced in the following list.

- Decisions portray the act of determining an output from a number of inputs, using logic defining how the output is retrieved from the inputs. Decisions may reference one or more knowledge models. Graphically, a decision is depicted as a rectangle, labeled with the decision name.
- Business knowledge models denote functions that encapsulate business knowledge, such as business rules, decision tables, or analytic models. They are illustrated as rectangles with clipped corners.
- Input data represent information used as an input by one or more decisions.
   When enclosed within a knowledge model, input data represent the parameters to the knowledge model. Graphically, they are depicted as ovals.
- Knowledge sources identify an authority for a business knowledge model or a decision, such as a person responsible for managing the represented knowledge, or a published document enclosing it. They are depicted as a shape having three straight sides and a wavy one.

Three kind of dependencies are defined between the presented elements, in the form of requirements. An *information requirement*, depicted as a simple arrow, represents any input data (or decision output) that is used as input to a decision. Graphically, an information requirement is depicted as a solid-line arrow with a filled arrowhead. A *knowledge requirement* denotes the invocation of a business knowledge model by a decision or by another knowledge model. Graphically, it is a dashed-line arrow, having an open arrowhead. Finally, an *authority requirement* depicts the dependence of a DRD element on another DRD element that acts as a source of guidance or knowledge. This may denote the fact that a business knowledge model is consistent with a published document, or that a specific person is responsible for a certain decision. Graphically, it is represented by a dashed line with a filled circular head.

Decision logic completes the (executable) description of decisions, by allowing one to define them in sufficient detail to support validation and/or automation. In general, decision logic can be modeled in several ways, which translate business expertise into business rules, analytic models, or other formalisms. DMN defines different kinds of value expressions to associate decision logic with elements of a DRG.

A *literal expression* represents decision logic as text that describes how an output value is derived from a set of inputs. Literal expressions may be represented with a formal, and possibly executable language, or they can be written in plain English. Formal expressions can be specified with the friendly enough expression language (FEEL) [21], which translates if/then/else data structures and calculations into executable expressions.

A *decision table* is a tabular representation of a set of input and output values, organized into rules that describe how a given input relates to one or more corresponding outputs [21]. The simplified FEEL (S-FEEL) is a subset of FEEL, used for defining expressions in decision tables.

An *invocation* is a tabular representation of how decision logic, represented by a business knowledge model is invoked by a decision, or by another business knowledge model.

Tabular representations of decision logic are called *boxed expressions*.

As an example, let us consider the 6-min walk test, introduced in Section 2.1. In some clinical situations, this test may be contraindicated for patients considered at risk. Relative contraindications are evaluated by a trained technician right before test execution. These are a resting heart rate of more than 120 bpm, a systolic blood pressure of more than 180 mmHg, and a diastolic blood pressure of more than 100 mmHg [24]. The technician performing the test must measure the patient vital parameters and decide whether the test can be conducted safely.

A simple DRD representing the decision about the conduct of the 6MWT is shown in Fig. 2.

The main decision regarding the Conduct of 6MWT is based on the assessment of Relative Contraindications, which are encoded as rules within a *business knowledge model*. This model is invoked by the main decision, which passes *input data* Arterial Blood Pressure and Heart rate as parameters for the model. Finally, the trained technician is indicated as a *knowledge source*, to represent the authority responsible for deciding about the conduct of 6MWT. The described DRD of Fig. 2 is linked to task evaluate relative contraindications of the BPMN process of Fig. 1. In general, there may be a one-to-one correspondence between a BPMN process task and a DMN decision. Yet, this is not mandatory, as the same decision can refer to multiple tasks and can be used across processes.

Figure 3 depicts the decision logic corresponding to business knowledge model Relative Contraindications, by using a decision table. In Fig. 3, input and output values are arranged into columns, while rules correspond to rows. At the top left corner of a decision table, a couple of attributes can be present, one to denote the kind of hit policy for the table, the other to indicate its completeness. A hit policy specifies how overlapping rules are handled, that is, how output values must be returned in case more than one rule match input data. In the presented case, character **U** is used to denote a *unique* hit policy for the table, meaning that no overlapping rule is allowed, that is, all rules are disjoint and only a single rule can be matched [21]. A completeness indicator can be optionally used to specify whether at least one rule in the table



Fig. 2 Decision requirement diagram representing the decision for the conduct of the 6-min walking test, with related input data and invoked business knowledge model Relative Contraindications

	► U	Conduct of 6MWT					
HIT POI		Arterial Blood Pressure. Diastolic (mm Hg)	Arterial Blood Pressure. Systolic (mm Hg)	Heart rate (bpm)	Conduct 6MWT (Boolean)		
RULES	1	-	-	> 120	False		
	2	> 100	> 180	-	False		
	3	> 100	> 180	>120	False		
	4	<= 100	<=180	<=120	True		
		×	INPUTS	*	OUTPUTS		

Fig. 3 Labeled decision table portraying the decision logic underlying decision conduct of 6MWT of Fig. 2

matches a possible input configuration. Symbol **C** denotes a complete table (DMN default), whereas **I** is used for incomplete ones. Finally, symbol "-" is used to denote that the value of a certain input is *irrelevant* for the final decision, that is, the decision can be made even if that input value is unknown.

#### 2.3 On Linking BPMN and DMN Diagrams

Whereas operational processes have received plentiful attention, decision modeling in business process management has been scarcely addressed to date. However, decision-making constitutes an important aspect of knowledge representation in business processes.

BPMN and DMN standards address decision-making from different and complementary perspectives. On the one hand, business process diagrams allow one to identify specific decision activities, which are usually captured as *user* or *business rule* tasks.

User tasks are defined as typical workflow activities, which are assigned to a human performer through the use of worklists and are executed with the assistance of a software application. Instead, business rule tasks provide a mechanism for the process to interact with a business rule engine for decision management [15]. On the other hand, DMN defines which are the decisions made in process tasks, which are their inter-relationships and their requirements for decision logic. DMN can be used independently from BPMN to model a domain of decision-making without any reference to business processes. The two standards are intended to be used according to the separation of concerns paradigm [22, 32], that is, by allowing designers to keep decision logic in a dedicated decision model and process logic in a process diagram. Decoupling process and decision diagrams can also be advantageous in contexts where process and decision managers have different roles and expertise. Indeed, despite allowing the use of gateways for drawing multi-criteria decisions, BPMN is not suitable for decision-design for many reasons. First of all, a process gateway is not entitled to properly "decide," but it rather specifies which path is taken

by a process instance, based on existing process data or event occurrence. Secondly, decisions in BPMN are often drowned into complex and hardly-readable diagrams and their formalization can become quite cumbersome [30]. Last but not least, slight changes of decision logic may have a strong impact on the whole process diagram, thus requiring substantial re-design.

DMN compensates BPMN by supporting the design of both operational decisions, which are made in day-to-day business processes, and strategic decisions, which are typically more complex and harder to represent. The easiest way to connect a DMN model to one or more BPMN processes is by allocating different kinds of decision-making to separate process tasks [21]. According to its level of automation, a decision can be discerned into human or automated. Human decisions are performed by organization personnel and are usually described informally at a high level, using natural language rather than decision logic. Conversely, automated decisions require that decision logic is formally specified. For these decisions, it should also be possible to validate that a process uses the results of decisions only in tasks or gateways following a decision activity [21]. In healthcare, partial automation is more common, since, despite being formalizable into automatic rules, most decisions are reviewed or validated by personnel.

A typical example of partially automated decision is drug dosing based on a patient's age and weight. This is performed by a physician, who takes into account some fixed thresholds suggested by clinical indications. According to [21], human decision-making can be enclosed in BPMN user tasks, whereas (partially) automated decision making can be allocated to business rule tasks.

# **3** The Methodological Framework

In this section, we present a new methodological framework for supporting the seamless design, implementation, and enactment of healthcare processes and decisions. We also tackle a few crucial aspects related to the management of information relevant for the execution of both processes and decisions. In particular, the methodology provides both a classification of the disparate facets of process-related decisionmaking and a qualitative analysis of the kinds of data involved in the various phases of process development.

#### 3.1 Overview

The proposed methodological framework, which extends the methodology introduced in [10], is a general-purpose approach grounded on standard BPM models and tools, but it is presented with special attention to the requirements related to the management of care pathways. In particular, we present a structured approach to delineate the main phases of care pathway development and, for each phase, we define which are the techniques and tools to be used and how. In addition, we provide the reader with an overview of which are the most important challenges that can be encountered in each phase. Finally, we suggest possible connection points with existing formalisms suitable to be used in the healthcare domain.



Fig. 4 Main steps of the proposed methodological framework. The Design phase has been exhaustively applied to the COPD case study, as detailed in Section 4

The viewpoint of the methodology is intentionally process-oriented, as the macrosteps portrayed in the central part of Fig. 4 correspond to specific phases of the business process life-cycle. Namely, we considered DESIGN, IMPLEMENTATION, and ENACTMENT, the latter one encompassing also performance analysis [19]. These phases constitute the core of the approach and are connected to process-related information, which is outlined at the top of Fig. 4. The mentioned life-cycle phases are framed by a dashed line to highlight that there is an open interaction and a continuous exchange of knowledge and information between the phases of process development and the organizational environment. The advancement from one phase to the next one is shown in Fig. 4 by small light-blue triangles.

In real application environments, it is common to deal with incremental process development, followed by iterative refinements and often with agile approaches, i.e., non-completely planned and heavily involving final users, in the related software development [33]. For this reason, the depicted life-cycle has to be considered to be continuous. That is, the output of the enactment phase is re-used as an input for the design phase. This iterative re-thinking and organizational change of both the structure and the outcomes of a process is often referred to as process re-engineering. Business process re-engineering is commonly defined as the "radical re-design of one or more aspects of a business process," aimed to improve process performance and to implement organizational changes [34]. In Fig. 4, the RE-ENGINEERING of both processes and decision-making is represented by thick curved light-blue arrows. A big re-engineering arrow links the enactment and design phases, denoting the need for process and decision re-design, based on information retrieved from process execution and analysis. Similarly, two smaller curved arrows are used to represent local re-engineering steps, aimed at refining the design of process and decision diagrams, based either on insights provided by simulation or on new requirements. In healthcare, process change is often needed to improve compliance to standards statements or to integrate health information systems [35].

Here, we will show how the presented methodological approach for care pathways design and enactment, even if applied partially, allows IT experts and care providers to interact and address some complex challenges [1, 6, 36]. The methodology is directed to expert process modelers, that is, we assume that a team of process and IT experts, actively interacts with care providers and clinicians to deeply understand and analyze the healthcare context. Indeed, as process development is a time-consuming task that requires IT expertise, we do not expect clinicians to directly model and maintain the given care pathway.

Table 1 summarizes some important questions whose answers need to be collected by process experts when applying the introduced methodology for developing healthcare processes.

#### 3.2 Design: Modeling Processes and Decisions

Reasonably, the development of decision-intensive healthcare processes begins with a DESIGN phase. This is organized into three main steps: REQUIREMENTS ENGI-NEERING, MODELING, and SIMULATION, depicted in Fig. 4. With respect to process and decision design, during Requirements Analysis a specific organizational problem is studied, and process participants, final objectives, and performance goals are defined. The latter ones are usually determined with help of indicators, established by teams of clinicians and healthcare managers who meet regularly to assess process performance. Among the indicators to consider in care pathways, in recent years, health care quality indicators (HCQIs) have been proposed by Organization for Economic Cooperation and Development (OECD) state members, as a way to assess, monitor, and compare the quality of healthcare systems both at regional and national level [40]. As an example, cancer mortality rates and the number admissions for asthma are quite traditional quality indicators, to consider in related pathways. Requirements engineering usually comprehends feasibility studies and the analysis, definition, and specification of requirements, aimed at understanding the studied domain and the role of the modeled pathway in the organizational context. In the healthcare domain, process and decision requirements related to care pathways stem either from organizational documents, medical literature, clinical guidelines, and local care policies, or they are collected by process modelers through direct interaction with both medical professionals and care managers. Usually, interviews with participants are conducted and modelers are invited to observe how the organization operates.

MODELING embraces both processes, data and decisions, based on the previously identified requirements. Process modeling allows one to filter out the irrelevant complexities of a real scenario, while providing a picture of the considered healthcare environment, tailored to spot improvements needed for re-engineering. Process modeling begins with Process identification, aimed at establishing the main process activities, their execution order, and the chosen modeling granularity. During process identification, the modeler decides which and how many processes should be outlined, considering final objectives and possible implementation requirements. Besides, it is crucial to ensure that the outlined processes do not overlap and that the interactions between resources correspond to reality. To this end, BPMN

Macro-phase	Phase	Step	Design questions
Design	Requirements engineering	Requirements analysis	How does the healthcare organization operate? Which are the final goals (clinical ones? related to healthcare policies? related to monitoring quality of care?) and which is their value? Who are the clinical/healthcare final users of the modeled care pathway? Are there healthcare policies or documents to follow?
	Modeling	Process identification	Which and how many processes can be discerned? Are the processes covering clinical or healthcare-organizational aspects? Which is the best modeling granularity that fulfills the goals of care providers? How do resources interact?
		BPMN diagram	Which are the main process aspects and how can they be represented (roles, activities, etc.)? Are there recurrent/known patterns? [37–39] Which are the clinical/healthcare data that need to be available for the process to be executed? Where are they stored? How are they exchanged? Does the model require execution details?
		Decision identification	Which and how many decisions can be discerned? Are decisions opera- tional or strategic (i.e., strongly based on aggregated medical data)? Do decisions influence the process flow? How?
		DMN Diagram	Which data requirements are needed for decision-making? Are these data coming from EHR/EMRs? Are they specific to this activity? Who is responsible for making the decision? Does the decision need to be automated or is it made by a human resource? Do clinicians require direct support for this decision?
	Simulation	Process and decision simulation	Is the process behaving as expected? Which are the costs and the dura- tion of the process? Are the bottlenecks related to bad modeling or excessive resource consumption? What aspects of the process need improvement? Can new indicators be proposed to better monitor the healthcare process?

Table 1 Summary of important questions that may be answered during each phase of the methodology shown in Fig. 4

 $\underline{\textcircled{O}}$  Springer

Table 1     (continued)			
Macro-phase	Phase	Step	Design questions
Implementations	Automation	Process implementation	How is the process enriched with information for execution? How is the graphical model translated/enriched? Are there organization constraints on execution?
		Testing and deployment	Does the engine have access to all services, client, and invoked appli- cation needed by the process? How does the process interact with other systems? How are instances grouped or synchronized (e.g., by patient, physician, and so on)?
Enactment	Execution	Process and decision execution	Who executes process tasks? Which execution data are relevant for performance analysis?
	Performance analysis	Conformance analysis	Does the process conform to the expected objectives? Are there process instances that cause bottlenecks?
		Monitoring	Which is the current status of the process? Are there unexpected short- ages of resources? Is there the need of sending alerts or managing exceptions?

choreography and collaboration diagrams can be employed to explicitly represent message-based communication between multiple processes and resources [15]. Usually, at this point, process elements are still defined at a quite abstract level.

Conversely, step BPMN diagram deals with the design of a more detailed process diagram, by making use of process elements defined in the business process model and notation standard. BPMN diagrams are suitable to represent the process control flow with sufficient detail and expressiveness, at different degrees of structural complexity and support for automation [19]. In addition, BPMN partially tackles other aspects of process modeling, such as the exchange of data produced and consumed during process execution and the management of process roles and their interactions. In particular, BPMN allows the designer to specify different roles within a healthcare organization and to specify which activities have to be performed by different roles, through *pools* and *swimlanes*.

During this phase, process designers adapt the process model to better suit the final design goals and to meet the local requirements of the application context. Designers can exploit existing patterns for modeling control flow [37], data [38], and time [39], to capture complex process aspects.

Similarly, decisions play a crucial role in process development and enactment, and, thus, it is important to understand how they affect both the process flow and the organizational environment. Hence, decision modeling is carried out in parallel or in conjunction with process modeling. Decision identification deals with the discovery of which are the decisions that affect the process and its organizational landscape: During this phase, modelers need to understand which are the relationships between decision-making and the process flow, and which are the repercussions of decision outputs on the process flow.

A decision can be modeled by means of a DMN diagram, which addresses different aspects of decision-making. In particular, decision requirement diagrams represent the internal structure of a decision and its relationships with other decisional steps. When needed, decision logic can also be specified, by using boxed expressions or decision tables. Whenever decisions are explicitly represented in a BPMN model, a DMN diagram can be associated to those process activities that encompass decision-making. In Fig. 4, this connection is represented through a simple double-headed arrow that links BPMN diagram to DMN diagram. However, according to the principle of separation of concerns [22, 41], it is beneficial to keep process and decision logics separate, as these are usually supported by different stakeholders, having distinct goals and expertise. We distinguish two main types of decision:

- Operational decisions are usually expressed quantitatively, in terms of process variables that are used by process gateways to determine which path has to be chosen. Usually, operational decisions are represented in BPMN as decision tasks (i.e., a task of type user or business rule), whose result is used by a following gateway [30]. In the healthcare domain, operational decisions concern daily activities, such as drug dosing, identification of a specific vital parameter to measure, and scheduling of examinations.
- Non-operational decisions include more complex and strategic decision-making, usually affecting high level process objectives [42]. In healthcare, such decisions

are often expressed in the form of natural language statements, based on medical evidence or professionals' knowledge, and rely on qualitative assessment. In a BPMN diagram, such decisions can be represented by a single decisional task, or they can span over more process activities. Examples of complex non-operational decision-making are clinical diagnosis and therapy planning, taking into account different kinds of knowledge and, possibly, sub-decisions.

Usually, DMN is used for supporting operational decisions, and patterns have been proposed to ease the extraction of such decisions from the process controlflow [30]. As for decision logic, DMN supports the representation of decision tables and logic formalisms that can be easily mapped to the friendly enough expression language (FEEL). Decision tables can be constructed by using almost any principle, for instance by using the approach presented in [43] when decisions need to be extracted from textual guidelines. In addition, algorithms for analyzing the completeness, consistency and non-redundancy of DMN decision tables have been proposed in [44], and tools such as Signavio currently support the verification of completeness for decision tables containing numerical entries. Moreover, DRDs can be useful to provide a compact and explanatory representation of non-operational decisions. In detail, DRDs can be used for decision compliance and management, as they allow managers to understand which data must be used for decision-making, which are the required sub-decisions to be made, and who is responsible for them.

As third and final step of the design phase, in Fig. 4 process and decision simulation is connected to process and decision modeling by means of dashed arrows, meaning that, despite being specific for the design of care pathways (as discussed in detail in Section 3.4), useful, and cost-saving, it does not constitute a mandatory design step.

#### 3.3 Dealing with Process-Related Information and Knowledge

Process-related information is represented at the top of Fig. 4, and the relationships with the introduced phases of process and decision development are highlighted by solid lines, depicted with a rounded arrowhead. During requirements engineering and modeling, Domain knowledge is used to provide a clear description of the relevant aspects of the modeled reality, by considering peculiarities, critical aspects, and goals of a specific healthcare organization. For instance, a care pathway for treating patients affected by pneumonia must deal with medical knowledge related to the management and treatment of pulmonary diseases, bearing in mind that we are interested in data regarding the organizational aspects of the treatment rather than in clinical knowledge. Similarly, Reference data help to delineate the requirements for process compliance. Examples of such data are reference values for medical parameters, input values for medical instrumentation, as well as administrative data regarding personnel salaries and skills, required duration of certain procedures, and quality standards. Reference data are strongly related to domain knowledge and local organization policies.

Then, Operational data regarding process activities, roles, and the organizational structure are also gathered during requirements analysis. During process and decision modeling, input and output data for process activities and decisions must be identified, and their structure, usage, and availability must be defined. In general, operational data produced, consumed, and exchanged by process activities must be considered during the design of a BPMN diagram, either to set activity attributes or to define data artifacts. During process design, data-related aspects are usually expressed through *data stores* and *data objects*. As BPMN is closely related to UML, some extensions can be adopted to enrich the graphical representation of data objects and data stores, by adding to process representation the direct reference to the UML database attributes that are touched by readings and inserts/updates required by each task. By adopting this solution, it is possible to represent the direct connection that exists between the conceptual model of the process and the conceptual schema of the accessed database [45].

In healthcare, operational data are typically stored within health information systems. Similarly, Operational decisions are used in day-to-day operations and must be defined to ensure that their input and output data are used properly within the process. Process and decision modeling deals also with non-operational data that are used for decision support. This information, often referred to as Aggregated data, can be used to measure the performance of the process within a specific application context and to make strategic decisions during process execution. In healthcare, aggregated data are extracted from data warehouses or decision support systems, to be used during process modeling and re-engineering. A subset of both operational and non-operational data, used to identify which are the aspects of a process model that are interesting to be analyzed during process execution, is represented by Conformance data.

Finally, Process metrics and KPIs are used to quantify process and decision quality and productivity. Data-based key performance indicators (KPIs) are measurable values that assess the progress of care pathways toward the achievement of the organizational objectives.

In BPM, KPIs are defined to monitor process execution, evaluate the quality of process and decision outcomes, and quantify the overall progress toward organizational goal achievement [20].

In this work, depending on their scope, we discerned two kinds of pathway KPIs. *Local KPIs* correspond to process and decision metrics and are mostly used to measure activity execution times, costs, and resource consumption. We refer to such indicators as "local," as they are extracted/used directly from/by multiple instances of one process model. As an example, the expected mean time duration to complete a care pathway related to the management of patients with breast cancer requiring a following chemotherapy is a local KPI, as it is related to the specific pathways and to possible "local" features of patients, reacting in different ways to the given chemotherapies.

Conversely, *global KPIs* are used to evaluate the strategic value of the performed (decision) activities both quantitatively and qualitatively. Global KPIs can be directly re-used in clinical decision modeling, to improve decision quality, patient prognosis, and predictive analytics. Usually, global KPIs rely on aggregated data and their impact can affect several activities and processes. In addition, they can be used for organizational and re-engineering purposes that go beyond the scope of the defined processes. These indicators are rather based on clinical and healthcare data, and

provide a summarized and focused overview of the modeled domain. Global KPIs can have both a clinical or organizational character. Continuing with the example related to the oncological pathway, a global KPI could be the ratio between the number of patients concluding in a successfull way the pathway and those who stopped the pathway for some health related issue. Such kind of global KPI could be possibly used inside the pathway to finely tune chemotherapy cycles as well as outside the pathway to activate, for example, suitable prevention actions.

As already underlined, pathway KPIs are strictly related to healthcare quality indicators, as we will show in the application of the proposed methodology to COPD patient management. Indeed, HCQIs may correspond to some measure used to structure and evaluate the considered pathway. As an example, the outcome indicator coded as *13a* "The proportion of patients with estrogen receptor negative invasive carcinoma who received adjuvant chemotherapy, out of the total number of patients with the same diagnosis" is an acknowledged quality measure of the process related to breast cancer treatment [46]. This value could be both a result of a general care pathway representing the overall management of patients with breast cancer and an indicator used inside a more specific pathway for managing chemotherapies, to estimate the percentage of patients that have to be considered for the given pathway.

Although care pathway KPIs are used during process analysis or simulation, they are defined during requirements analysis and process identification, in order to better tailor the process model to the final goals.

#### 3.4 Integrating Simulation in the Design of Care Pathways

Process simulation is one of the most popular and widely used technique for quantitatively analyzing process models [20]. Simulation encompasses both the execution of a model under hypothetical conditions in a defined simulation environment and the analysis of the execution output [47].

During process design, Simulation can be used to support the quantitative analysis of pathway models. Specifically, Process and decision simulation promotes the study of the execution of modeled pathways and decisions in a realistic or ideal scenario, without resorting to pathway and decision implementation. Being faster and more flexible than process enactment, simulation tools and techniques foster pathway validation and provide reliable estimates of resource consumption and execution times. Indeed, process configuration and enactment may be highly demanding, while stakeholders are interested in obtaining prompt and reliable estimates of process behavior, associated costs, required resources, and quality of outcomes. Simulation allows one to validate a process behavior, by spotting bottlenecks, and by evaluating resources costs and consumption. Besides, the duration of the overall process and of single activities can be estimated and ideal or worst-case scenarios can be studied. Decision simulation is often used to reproduce the outcomes of decision tables given an arbitrary set of input values and to verify table completeness. As previously mentioned, simulation results can be re-used to promote process adaptation to expected performance requirements. Moreover, simulation can be used for what-if analysis to explore alternatives, eliminate inadequate re-engineering proposals, and refine current as-is procedures.

Beside the continuous support to designers during care pathway modeling, in the proposed methodology, we underline the importance of simulation for stakeholders who are responsible for decision-making within the healthcare organization. Indeed, simulation support healthcare decision makers in *resource allocation and management*. Through simulation, decision makers can verify whether the resources associated to a care pathway have been suitably considered, with respect to the estimated number of patients and with respect to the specified goals for the pathway (e.g., reach the end of the treatment within 30 days since the admission of the patient).

Both local and global pathway KPIs could be considered together with HCQIs to see their mutual interactions in a "what-if" analysis context. As an example, in the chemotherapy domain, the previously defined indicator 13a is expected to be at least 80% [46]. This value could be used to distinguish the percentage of patients following a path of activities related to chemotherapies from other patients receiving different treatments. When simulating the same care pathway, one could change the value of 13a to the target value of 90% in order to monitor the effects of such change on the local KPIs of the pathway, such as its overall cost or its average duration.

Moreover, simulation can provide healthcare decision makers with some *evaluation of HCQIs* related to the considered pathway. As an example, in the design of a care pathways for patients with acute stroke, we could consider whether the HCQI *Acute stroke mortality rate* is maintained low, according to the expected features of patients, the expected results of the designed activities, and so on. In this direction, simulation can be used to estimate, as a kind of decision-support what-if analysis how HCQIs change with respect to different scenarios. Finally, simulation results can also feed some indicators that either are employed in external processes or support strategic decision making, or are used to obtain more complex HCQIs.

#### 3.5 Implementation and Enactment

Following design, the methodological framework deals with Implementation, which can be realized in different ways, depending on which IT infrastructures are already available in the organizational environment. Implementation is realized in different steps, usually starting from the automation of single processes and, then, by realizing the coordination between them. When starting from a single process, Automation is usually the first step that is applied for implementation. Automation focuses on the selection of software and services for process execution, encompassing both implementation and deployment aspects. In Fig. 4, by Process implementation we intend mostly process configuration, that is, the integration of a process model with technical information that facilitates enactment. Then, the enriched BPMN diagram is mapped into a formalism suitable for execution. Processes are executed by engines, which link client applications, supports the distribution of work among different process actors, and coordinates activities [2]. Most existing process engines support the direct deployment of standard XML process descriptions, generated by modeling tools, without the need of further mappings.

Process automation is closely related to Testing and deployment. Traditional testing techniques from software engineering can be used to test the automated process, to see if it behaves as expected. During this phase, integration and performance are tested for detecting potential run-time problems. Finally, the system is deployed in its target environment and complementary actions, such as data migration and usertraining may be executed. In general, one advantage of using standard modeling languages is that process engines can easily interpret them and, within healthcare organizations, such systems may already be used by the administrative staff.

The final phase of the introduced methodological framework deals with Enactment, which consists of Execution and Performance analysis.

In detail, Process and decision execution encompasses the actual run time of a decision-intensive process. This is usually the result of the process engine interacting with client applications. Execution constraints must be observed, and execution exceptions arising from unexpected occurrences must be properly handled. In this methodology, we do not explicitly deal with exception-handling, as this topic depends on the kind of system used for implementation and solutions have already been proposed in the BPM community [48].

During and/or following process execution, performance analysis may be carried out to monitor enactment or to analyze how well the process is performing, compared to desired performance goals. In Fig. 4, the relationship between process execution and performance analysis is depicted by a double-headed dashed arrow, meaning that the latter may be optionally performed during or after process execution. Monitoring allows one visualizing the current status of process and activity execution, in order to be able to enact corrective measures if the process is not behaving as expected. During process monitoring, bottlenecks can also be detected.

Analysis tools and algorithms can also be used to verify specific properties, such as done for the satisfaction of temporal constraints in [49].

Conformance analysis is employed to verify the fitness and appropriateness of a process with a given pattern or organizational model, in order to check whether the process is well-incorporated the organizational practice. Both structural and behavioral aspects can be analyzed and the results of the analysis can be used for re-aligning the process model to reality. In general, multiple cycles of monitoring and conformance analysis are executed. In addition, formal methods, such as Petri nets, can be applied for analyzing process behavior [50].

During process enactment, conformance data and process metrics and KPIs are used to monitor the status of the running process and to analyze different executions. Usually, valuable execution data are gathered in the form of Execution logs, that is, long files that contain information about events that have occurred during process enactment. Examples of such data are starting and ending times of activities, task duration, role assignments, and message exchanges. Execution logs are analyzed for process re-engineering and alignment, or for discovery and mining purposes.

#### 3.6 Managing Temporalities of Care Pathways

Even though a deep discussion related to the management of temporal issues in care pathways is out of scope with respect to the content of this paper, in the following we will discuss the main temporal issues related to care pathways. Besides, we will show how the proposed methodology can be used to easily embed a suitable management of such time aspects. Temporal aspects of care pathways are usually complex and need to be suitably modeled, both in Requirement, and Modeling phases. As an example, in modeling preliminary stages for the 6-min walk test in Section 2, we need to represent that the patient as to rest for at least 10 min before executing the test. Further temporal constraints among different activities may also need to be represented. For instance, it could be interesting to show that the efficacy of a therapy must be evaluated 2 weeks after the treatment has been prescribed, as in the management of COPD that will be later discussed. Furthermore, periodicities need to be considered, since some carerelated activities are inherently periodic, as, for example, the assessment of some chronic pathology or the administration of a given drug.

The representation of temporal constraints in BPM systems have been considered in literature, also considering clinical processes [9, 51–54]. In general, the focus of such proposals is both (i) to design temporal constraints in BP models and (ii) to verify some basic properties of temporal BP models. In particular, the concept of *dynamic controllability* has been introduced and extensively studied. Suitable algorithms have been proposed to verify at design time whether a process model can be executed while satisfying all the given temporal constraints and allowing activities performers to take any possible duration within the range of the allowed ones [9, 53]. The proposed solutions can be applied also to suitably extended BPMN models, as they use the most common BP constructs, as activity, subprocess, gateways, and so on. Thus, from this point of view, the proposed methodology is well suited to embed tools and models for checking temporal features of BP models.

Recently, modular BPMN-based temporal patterns have also been proposed in [55, 56], explicitly focusing on clinical tasks. The advantage of these BPMN patterns is that temporal aspects are suitably represented and embedded within a sub-process that can be edited by modelers and executed directly on a process engine. Thus, such patterns can be seamlessly and directly specified within BPMN care pathways and suitably dealt with in the proposed methodology.

Moving to the Simulation and Execution of time-aware care pathways, two different aspects arise: The first one is related to updating temporal constraints according to previously executed process tasks; the second one deals with the runtime management of possible constraint violations and of the related reaction policies. As shown in [11, 55] for some basic patterns, designers can leverage exception handling of BPMN 2.0 to specify suitable patterns for supporting different kinds of reactions to constraints violation. As an example, if a therapy duration exceeds its maximum duration, the model designer could specify either that such therapy must be immediately interrupted (strong policy) and some other compensation actions could be enacted or that the therapy will continue (weak policy), but some other actions have to be done (sending a warning to the physician) [11].

As a final comment, it is worth noting that powerful solutions to support timerelated aspects of care pathways through BPMN are mainly based on the expressive power provided by the use of events, event-related activities, and exception-handling mechanisms. As an example, timer events can be used to specify either a specific delay from their activation or a certain time-date or a specific period of time within the process model, to trigger some specific clinical activity or some conditional path in the control flow.

# 4 Design and Simulation of Decision-Intensive Care Pathways for Chronic Obstructive Pulmonary Disease

In this section, we apply the introduced methodology to a real healthcare scenario, in order to discuss modeling choices and design challenges, and share the benefits brought by the proposed approach in a real clinical context.

The methodology discussed in the previous section has been applied to support the design of diagnostic-therapeutic care pathways for the management of chronic obstructive pulmonary disease in the region of Veneto, in northern Italy. The process and decision diagrams have been designed to provide a compact representation of healthcare procedures aimed at easing the understanding and improvement of as-is healthcare pathways and of their organizational aspects. One of the main modeling goals was the detection of pathway flaws that led to undesired, yet reducible healthcare expenses. Healthcare pathway implementation and enactment are out of the scope of this paper that mostly focuses on care pathway design and simulation. Hence, here we will consider in detail the design of COPD-related care pathways.

Requirements have been collected by analyzing textual documents authored by the Regional Healthcare System of Veneto, including both regulatory standards and organizational plans, and clinical practice references. In addition, we directly interacted with the regional board, composed of care managers and clinicians responsible for the actuation of the care pathways, and we were directly observing how care interventions have been carried out in a few regional hospitals. Finally, standard American and European guidelines for COPD management have been studied and used for process compliance [23, 57].

Among existing tools for process design and management, we chose the Signavio Business Transformation Platform, which integrates process modeling, decision design, and simulation features for cost analysis [18]. The web-based Signavio Process Editor has been used for process design and simulation, as it guarantees full adherence to the BPMN standard and supports different kinds of process simulation and cost management on several user-defined scenarios. In addition, it provides several functionalities that ease collaborative design, diagram reviewing, and commenting. The Signavio Decision Manager has been chosen to model DMN diagrams and to simulate decision table execution.

In the remainder of this section, we introduce the modeled clinical domain in Section 4.1 and we explain the design of both processes and decisions with BPMN and DMN in Sections 4.2 and 4.3. Then, we discuss the re-use of information through indicators in Section 4.4 and present simulation results in Section 4.5.

# 4.1 Chronic Obstructive Pulmonary Disease

The World's Health Organization defines COPD as "a life-threatening lung disease characterized by chronic and not fully reversible obstruction of lung airflow, which interferes with normal breathing" [58]. Despite being both preventable and treatable, COPD global prevalence and socioeconomic impact remain significantly high [59], thus making the disease one of the leading causes of disability and death in developed countries.

Whereas COPD is often underdiagnosed in younger patients, as respiratory abnormalities become noticeable only when internal organ damage is already advanced [60], current diagnostic criteria may lead to over-diagnosis in the elderly [61]. A major proportion of COPD burden is represented by exacerbations, i.e., acute and unexpected worsening of respiratory symptoms that require a change in management.

In order to improve the management of COPD, integrated diagnostic and therapeutic care pathways have been proposed at a regional scale, with the intent of improving the effectiveness of COPD treatment, the prevention of exacerbations, and the coordination between different care providers. The development of such care plans aims to standardize the critical steps of long-term patient treatment and the overall decision process. In particular, it is crucial to clarify which is the role of each involved resource, which are the responsibilities, and which is the optimal timing for intervention. Besides, clinical, social, educational, and organizational objectives are set by considering economical expenses and resource availability.

The main goal of COPD diagnostic and therapeutic care pathways is the improvement of primary care quality mostly by reducing costs, intended both as patient lives and economical burden, related to misdiagnosis and to the prescription of unnecessary medical exams and treatments. The definition of such plans also encompasses a set of indicators that are defined to evaluate disparate aspects of the healthcare process, such as its structure, clinical and organizational efficiency, performance, progress toward the set final objectives, and overall economic impact. Prevention measures, educational programs, and counseling services are also included in the care plan, with the aim of training patients to recognize symptoms worsening and to improve the self-management of the disease in home care settings.

The proposed methodology supported regional care stakeholders to design COPDrelated decision-intensive care pathways. Data used for the design and simulation are based on official regional reports [62, 63] and on other bibliographic resources provided by healthcare stakeholders [64–66]. Previous research on the burden of COPD in Italy has been presented in [64]. Data regarding COPD expenses have been considered with respect to those obtained by the interviews with regional healthcare stakeholders, although the number of patients affected by COPD in Veneto is increased from 6.2% of 2003 to 8.5% 2015. The major sources of expenses are hospitalization and oxygen therapy. The problems highlighted in [64] are similar to those highlighted by the experts that worked on the care pathways: poor results of smoking cessation programs, limited use and access of spirometers in primary care, underdiagnosis of COPD.

Such design phase is an initial step that should allow to reach the definition of a region-wide standard management of COPD patients, after a deep re-engineering phase. Healthcare stakeholders have been involved in design phase of the methodology, with an agile approach. Healthcare stakeholders participated to the design together with BPMN designers, who asked questions highlighted in Table 1 to them during the design meetings and collected answers in working and informal documents. BPMN and DMN documents, integrated with comments, together with simulation descriptions, are the formal and final results of this iterative design phase.

The modeled BPMN processes have provided healthcare professionals with a comprehensive and accurate overview of the COPD care pathways. After applying

the proposed methodology, improvements in common understanding and cooperation have been noticed. In earlier times, each resource used to work independently and this fragmentation resulted in longer waiting times for patients, miscommunication, and information overload. The standard definition of the care processes has served to delineate everybody's responsibility, while improving information exchange, cooperation between resources, and coordination between several processes. Besides, from a technical point of view, the process model has been used for validating the consistency and efficiency of the proposed solutions, thus constituting a sound trace for implementation.

#### 4.2 Modeling COPD Pathways with BPMN

In the assessment and management of patients with COPD, three main intervention steps have been discerned, namely (i) *COPD diagnosis and assessment*, (ii) *ordinary management of stable COPD*, and (iii) *management of COPD exacerbations*. These steps are described in medical literature [23, 57], and are also be identified in the asis processes that are currently applied in Veneto. For each one of the aforementioned steps, a BPMN process diagram has been designed, following the principles of well-structured process design to lower diagram complexity and reduce the probability of committing modeling errors [28, 29].

The three phases included in the studied care pathways for the management of COPD are depicted in the BPMN process diagram of Fig. 5, each instance of which represents the treatment of a single patient. Each one of these aspects of COPD management is represented as a collapsed sub-process, which relates to the other process elements as described in the following paragraph.

The overall care process begins in primary care with the COPD diagnosis and assessment, during which practitioners, organized in primary care units (PCUs), have to detect the disease and classify the patient according to the most pertinent severity level. This assessment phase is carried out only once and, then, any patient diagnosed with COPD enters a multidisciplinary program for monitoring and managing the disease, which is enacted periodically, based on the assessed COPD stage. The studied healthcare context of Veneto provides a staging system based on three main levels of severity, which is slightly different from the one presented in [23]. In particular, all patients having a spirometry-related value FEV<sub>1</sub>  $\leq$  50% are considered at high risk and belong to stage 3 (i.e., there is no stage 4). The general practitioner is in charge of COPD diagnosis and of monitoring and validating the decisions made by pulmonary specialists.

General and pulmonary examinations considered in the Management of stable COPD are scheduled based on the severity of the disease. For this reason, the subprocess representing the ordinary management of COPD is executed periodically. In BPMN, this periodicity is expressed by setting a looping condition for the whole Management of Stable COPD sub-process. In Fig. 5, this is highlighted by a standard loop marker [15], depicted at the bottom of the sub-process.

However, during ordinary COPD management, exacerbations can occur. In the BPMN process of Fig. 5, the occurrence of an exacerbation is represented by means of non-interrupting boundary conditional event Exacerbation episode, which is triggered when an exacerbation occurs.

Non-interrupting boundary events allow the activity to which they are attached to remain active and a new a token is generated to drive a new flow, parallel to the continuing execution of the activity.

Therefore, as event Exacerbation episode is non-interrupting, Management of COPD exacerbations starts while ordinary Management of stable COPD is still being performed. Exacerbations must be promptly treated in order to reduce the impact of the current worsening and prevent the development of future episodes. In general, the treatment of exacerbations does not affect the intervention planned for the ordinary management of COPD.

However, ordinary medical inspections must be re-scheduled whenever the stage of the disease changes, either due to the progression of the disease or to an exacerbation episode. In Fig. 5, this is represented by boundary interrupting signal event Re-schedule, which is triggered by a corresponding signal event thrown either within sub-process Management of stable COPD or following conditional event Stage changed, which captures a change of stage due to an exacerbation. Signal events are used for broadcast communication and, in the depicted process, they represent the



Fig. 5 BPMN diagram outlining the three main steps of the studied COPD care pathway, each one represented as a sub-process. Sub-process COPD diagnosis and assessment is always executed first, followed by Management of stable COPD, which is performed periodically, according to the assessed stage. Ordinary examinations can be rescheduled whenever the COPD stage assigned to the patient changes due to the progressive worsening of the disease or following an exacerbation episode, the latter handled by sub-process Management of COPD exacerbations

interaction between the two sub-processes Management of stable COPD and Management of COPD exacerbations. Throwing signal events are characterized by a filled triangle marker, whereas catching signal events have a unfilled maker of the same sort. Whenever signal event Re-schedule is caught, sub-process Management of stable COPD is interrupted and restarted, according to the periodicity defined for the newly assessed stage. The end of the described macro-process is reached once all interventions for COPD management and exacerbations are concluded.

Figures 6, 7, and 8 show the BPMN diagrams representing the structure of the three introduced sub-processes. The detailed discussion on the structure of such sub-processes and of the related clinical activities is reported in Appendix A. In these process diagrams, different tasks types are used to denote decision tasks, i.e., process activities incorporating decision-making, which can be represented through a linked DMN decision model, at different levels of detail. As described in Section 2.3, we employ *user* tasks to denote activities that encompass human-decision making, whereas *business rule* tasks are used to highlight partially automated decision-making [21].

All the other process activities are represented as *abstract* tasks or sub-processes, that is, generic activities whose behavior is not further specified [15].

In the BPMN diagrams, the recording of data to persistent databases is represented through data stores. For readability reasons, we used data objects to show only the exchange of information between different resources, while omitting details related to data flowing through activities executed by the same resource.

In particular, the presented processes need to communicate with the following information systems: The primary care information system, represented by data store PCU db, is the system used by general practitioner to store patient data – This system is usually shared among the practitioners that belong to the same PCU; The hospital information system, represented by data store Hospital db, stores different patient medical records; Finally, the health district authority information system, represented by data store HC district db is mostly used to record data regarding vaccinations and related campaigns. Nowadays, the three information systems are only partially connected, since data about drug prescriptions and examination scheduling are shared between PCUs and hospitals, but most data are still exchanged between professionals through reporting documents. The region of Veneto is currently working on developing a unified electronic health record that will allow practitioners, hospital personnel, health district authorities, and pharmacies to access the same data repositories.

#### 4.3 Modeling Clinical Decisions with DMN

Process-related decisions have been modeled in detail by using DMN diagrams and, sometimes, by specifying more detailed decision logic. A few decision models corresponding to the COPD diagnosis and assessment process had already been introduced in [10]. In this paragraph, we focus on describing the decision modeling for the previously described management of COPD exacerbations process.

In the management of COPD exacerbations, a crucial step is represented by the diagnosis and assessment of an exacerbation, which is evaluated by combining physical findings with the clinical history of the patient [57, 67]. In general, clinical diagnosis is probably the most important example of complex, decision-intensive



**Fig. 6** BPMN process diagram for COPD diagnosis and assessment. The main decision activities identified with the help of experts are case history and working diagnosis, which is a preliminary step of decision Diagnosis of COPD, and COPD staging

medical task. Indeed, diagnosis encompasses aspects of human decision-making that rely on medical knowledge, professional experience, and sound sources of clinical information. For this reason, modeling clinical diagnosis by means of formal approaches can become quite cumbersome, as it involves complex (temporal) reasoning and uncertain and incomplete knowledge [68]. Nevertheless, the representation of



Fig. 7 BPMN process diagram for the ordinary management of stable COPD

structure of a decision and of its meaning is interesting for the organization and can be realized by using DMN decision requirement diagrams, which allow one to describe decision-making at quite a high level.



Fig. 8 BPMN process diagram for the extraordinary management of COPD exacerbations

As an example, consider the DMN diagram representing the diagnosis of a COPD exacerbation, depicted in Fig. 9. The diagnosis of an exacerbation relies on the clinical presentation of the patient, who consults the general practitioner complaining of substantial symptomatic and physiologic deterioration [23].



Fig. 9 Decision requirement diagram representing the main decision Diagnosis of COPD exacerbation, together with the input data needed for making the decision, the used business knowledge, and the sources responsible for the decision. The decision is linked to task Diagnosis of exacerbation of Fig. 8

In the DRD of Fig. 9, Diagnosis of COPD exacerbation is portrayed as the main decision, which ideally responds to the question "Is the patient experiencing an exacerbation of COPD?". The decision is made by a general practitioner, represented as a knowledge source, who is responsible for assessing the degree of symptom worsening and for evaluating if the exacerbation is life-threatening. Several clinical findings have to be considered when assessing an exacerbation. During a physical examination, pulse oximetry is performed to assess the need for supplementary oxygen therapy. Besides, hemodynamic stability, use of accessory respiratory muscles, andresponse to treatment are evaluated. The general practitioner must also consider the number of previous exacerbation episodes and the presence of comorbidities, which are combined with information about COPD severity to predict the patient risk of exacerbations. COPD severity corresponds to the output of sub-decision Evaluate severity of COPD, which is based on spirometric assessment. Results of previously administered CAT, spirometric assessment, and history of exacerbations are retrieved from the PCU db, depicted as knowledge source. Probability of comorbidities is a business knowledge model that describes the kind of comorbidities that are commonly found during an exacerbation and the probability of finding more of them. Business knowledge models describe a reusable fragment of decision logic or knowhow, which can be retrieved by guidelines or by local organizational knowledge. In the presented context, the knowledge is based on data collected in the region, since the probability of comorbidities changes according to the distribution of the

Probability of Comorbidities
Common comorbidities associated with poor prognosis of COPD
exacerbations are: congestive heart failure, coronary artery
disease, renal and liver failure.
According to data gathered in the Region of Veneto, the percentage
of patients presenting comorbidities is as follows:
12% of COPD patients has only one comorbid condition;
21,1% of COPD patients has two comorbid conditions;
22,6% of COPD patients has three comorbid conditions;
16,2% of COPD patients has four comorbid conditions;
The rest of patients can have from 5 to 10 comorbid conditions.

Fig. 10 Boxed literal expression, written in plain English, and representing the decision logic of the Probability of comorbidities business knowledge model

population in the territory. Similarly, business knowledge model Assess risk of exacerbations encloses the knowledge required for assessing exacerbation risk, based on the combination of COPD severity, symptomatic assessment and history of previous exacerbations [23].

The diagram of Fig. 9 is linked to the user task Diagnosis of exacerbation of the process shown in Fig. 8. The result of the decision is used by gateway Is exacerbation life threatening to determine if the patient must be treated in outpatient settings or hospitalized.

The DRD introduced in Fig. 9 encompasses both human decision-making and partially automated decisions, the latter ones reviewed and validated by the general practitioner. For the main decision Diagnosis of exacerbation, we decided not to specify decision logic, as the presented DRD suffices in explaining the expected structure and input data needed to make this human decision. However, a few elements in the considered diagram have a decision logic specified at different levels of abstraction, by using the various expressions provided by the DMN standard [21]. In detail, decision logic for business knowledge model Probability of comorbidities is represented by a boxed a literal expression, shown in Fig. 10, written as plain English text and, thus, not suitable for automation. Conversely, sub-decision Evaluate severity of COPD is based on well-defined rules that determine how the amount airflow

E	Evaluate Severity of COPD					
E١	Evaluate Severity of COPD Table					
FE	EV1		Spirometric assessment.FEV1			
FE	EV1/FVC		Spirometric assessment.FEV1/FVC		FVC	
1	U	Evaluate Se	everity of	COPD Table		
		FEV 1/FVC (percentage)		FEV1 (percentage) {"N	COPD Severity	
   					{"Mild", "Moderate", "Severe"}	
	1	< 70 %		>= 80 %	"Mild"	
	2	< 70 %		[50%80%)	"Moderate"	
	3	< 70 %		< 50%	"Severe"	

Fig. 11 Decision table representing the decision logic for decision Evaluate severity of COPD, based on spirometric data

limitation should be associated to a specific COPD stage. Hence, decision logic can be easily represented by means of a decision table that relates spirometric values FEV1 and FEV1/FVC to the corresponding COPD stage. Figure 11 shows the decision logic of decision Evaluate severity of COPD, which invokes decision table Evaluate severity of COPD table, passing Spirometric assessment.FEV1 as FEV1 parameter and Spirometric assessment.FEV1/FVC as FEV1/FVC parameter. Decision evaluate COPD severity, invokes the corresponding decision table. The presented table has been verified with Signavio and, as expected, it is not complete as no rule exist for ( $\leq$  70%, *any*). However, this omission was intentional, since this latter case is not considered in COPD treatment.

Finally, as an example of formal expression allowed in DMN, consider the decision logic for business knowledge model Assess risk of exacerbations, represented in Fig. 12 through a boxed FEEL expression which describes how an output value is derived from its input values. The presented formal expression is used for combined COPD assessment in order to understand which is the impact of COPD on an individual patient, in terms of future exacerbations, based on current symptoms, spirometric assessment, and exacerbation history [23]. Patients are classified into four main categories, which summarize the risk of exacerbations and expected health status. For instance, a patient with a CAT score of 16, assessed with "Moderate" COPD, and a history of three exacerbations within the last 12 months has a risk of exacerbations corresponding to "High, more symptoms".

#### 4.4 Dealing with Information and Process Indicators

The modeling of the described COPD processes and decisions underlined the strong connection that exists between processes and related clinical and organizational information, throughout all the design phase. On the one hand, data needed for process and decision execution must be identified during process design to define which is the information needed during process enactment. On the other hand, new informational value generated during process simulation and execution must

Risk of Exacerbations
If COPD Severity IN ("Mild", "Moderate") or
ExacerbationsPerYear IN ([01]) and CAT <= 10
then Risk of Exacerbations = "LOW, LESS SYMPTOMS"
else if COPD Severity IN ("Mild", "Moderate") or
ExacerbationsPerYear IN ([01]) and CAT >= 10
<b>then</b> Risk of Exacerbations = "LOW, MORE SYMPTOMS"
else if COPD Severity = "Severe" or
ExacerbationsPerYear IN ([01]) and CAT <= 10
<pre>then Risk of Exacerbations = "HIGH, LESS SYMPTOMS" else Risk of Exacerbations = "HIGH, MORE SYMPTOMS"</pre>

Fig. 12 Decision logic for the Assess risk of exacerbations business knowledge model, adapted from guidelines on combined COPD assessment [23]

be properly managed and used for re-engineering as-is processes and decisions. In the management of chronic diseases, most clinical knowledge and conformance data stem from standard guidelines, whereas operational data are retrieved from process execution logs or health information systems [69]. However, health information systems are usually meant to serve multiple and disparate care processes and, thus, data availability, standardization, and maintainability may become a complex issue.

In the studied context, a first step toward the identification of a minimum set of data that are relevant for process execution and audit is represented by the definition of process indicators, used to assess the quality of the provided care from different viewpoints and levels of abstraction.

The definition of indicators is part of the care process specification and is done by a "focus group" of clinical experts, who identify clinical, organizational, and social criteria for evaluating process outcomes [70]. Indicators are used to validate the structure of the process and to evaluate the organization potential in terms of resource distribution and consumption. In addition, they are useful to estimate process compliance with respect to both clinical and organizational outcomes, and cost sustainability.

#### 4.4.1 Local KPIs

An example of local KPI related to process metrics is the average total cost spent by the regional healthcare system for treating COPD patients assessed with stage 3. Local KPIs allow one to quantify the probability that a process execution path is preferred over another, within the context of a data-based process gateway. For instance, the percentage of smoking patients determines how many times the care providers involved in the process of COPD treatment have to deal with smoke cessation. In the region of Veneto, it has been estimated that the probability of contracting COPD for smokers are 25% higher than for non-smoking patients, and can increase up to 30% in patients that smoke more than 30 cigarettes per day. In the process of Fig. 6, this probability value influences the execution of all tasks that concern the administration of smoking questionnaires, as Mondor and Fagerstörm tests can be skipped if the patient is not a smoker. Local KPIs can also be derived and evaluated with process simulation techniques, as explained in Section 4.5.

#### 4.4.2 Global KPIs

The indicators defined by the regional healthcare system of Veneto are global KPIs used to give a high level estimate of the impact of care standardization in terms of costs, patients involvement, and medical personnel competence. Such indicators have been highlighted as text annotations in the process diagrams of Section 4.2. Indicators were provided by clinicians and they have been defined by the regional care stakeholders. Our goal was not to define them directly, but rather to show where in the process they are calculated.

For instance, Indicator 5: Performed vaccinations is defined to quantify the number of performed influenza vaccinations, in order to assess the quality of secondary prevention of COPD. In the documents provided by the Veneto region, it is defined as follows.

**Definition 1** (Indicator 5—Performed vaccinations (Fig. 7))

**Description:** Secondary prevention of patients diagnosed with COPD.

- **Numerator:** Number of people aged  $\geq 40$  diagnosed with COPD (i.e., ICD9-CM 491.2x and 496.x) and having received a vaccination against influenza in the last 12 months.
- **Denominator:** Totality of patients aged  $\geq$  40 diagnosed with COPD (i.e., ICD9-CM 491.2x and 496.x).
- **Needed Data:** Patient ID, diagnosis of COPD (ICD9-CM 491.2x and 496.x), influenza vaccination, date of last vaccination.

Indicator 5 is associated to task Verify administered vaccinations of the process shown in Fig. 7 and experts expect the value of this indicator to remain greater than the 40% of all the patients diagnosed with COPD (and quantified by Indicator 2A). Any value lower than the expected threshold should be considered an index of poor process performance, to be further investigated by considering re-engineering. With process management tools, Indicator 5 can be calculated by counting how many times the process branch labeled with YES is followed after each exclusive gateway Vaccination administrated? As for data requirements, this indicator suggests that the following data need to be collected during the care process: A patient identifying code, the diagnosis of COPD registered according to the ICD-9-CM coding systems, the influenza vaccination, and the date of the vaccination. As for the global scope of the indicator, data about the performed vaccinations gathered during the management of stable COPD in Veneto, are combined with data collected at a national level to evaluate the efficacy of routine vaccination programs. In the years 2002–2009, 32% of Italian population received vaccination against influenza, whereas in the years 2010-2016, the same value dropped to 18%. Among patients with COPD, around 75% of them received a vaccination against influenza in the same time-span. This drop in the number of vaccinations, highlighted by Indicator 5, can be explained by the recent spread of several anti-vaccination movements that developed in the considered territory in the last years. As a consequence, new vaccination programs were introduced, patients were actively informed about the risks of non-vaccination, and a service for delivering vaccinations at home was provided. This latter intervention was directed not only to COPD patients, but to all those considered at risk by the National Healthcare System, whence the global scope.

As a further example, here we report the definition of Indicator 7, translated in English from Regional documents related to the care pathway.

**Definition 2** (Indicator 7—Appropriateness of therapy (Fig. 7))

**Description:** Appropriate therapeutic use of only bronchodilators in people diagnosed with COPD in stage 2.

- **Numerator:** Number of people diagnosed with COPD (i.e., ICD9-CM 491.2x and 496.x) and assessed with stage 2 and receiving only bronchodilator therapy.
- **Denominator:** Totality of patients diagnosed with COPD (i.e., ICD9-CM 491.2x and 496.x) and assessed with stage 2.
- **Needed Data:** Patient ID, prescription data regarding short and long term bronchodilators (prescription code ATC R03).

The remaining indicators defined within the processes can be used as Indicator 5 and Indicator 7 to define (clinical) data requirements and measure clinical and organizational outcomes.

As for relationships between local and global KPIs and HQCIs in the specific case of COPD care, we mention here the quality indicator *PQI 05 - Chronic obstructive pulmonary disease (COPD) or asthma in older adults admission rate* [71], which is closely related to the definition of COPD patients, adopted by the Veneto Region and related to Indicator 1—prevalence of patients with suspected COPD. Indeed, the identification of patients with COPD is based on the following criteria from hospital and drug prescriptions:

- Admissions with a principal/secondary diagnosis of COPD (codes ICD-9-CM: 490-492, 494, 496), ages 45 years and older, or
- Sound and continuous drug prescriptions related to respiratory treatments (the detailed specification is done with respect to the overall duration of the therapy in the last two years, the number of prescriptions, and the ATC therapeutic class of drugs).

A further example of HQCI that could rely on the modeled COPD process is the *Indicator NQMC:010510* defined as *Use of spirometry testing in the assessment and diagnosis of COPD: percentage of members 40 years of age and older with a new diagnosis of COPD or newly active COPD, who received appropriate spirometry testing to confirm the diagnosis* [72]. Indeed, such indicator is enforced to be 100% for patients treated according to the given pathways (without considering some small differences with respect to the age of patients), as all patients undergo spirometry either in PCUs or at the hospital.

#### 4.5 Simulation of COPD Processes

In the application of the proposed methodology, we used simulation for validating the behavior of the modeled COPD processes by spotting pitfalls related to erroneous control flow. Then, we evaluated the performance of the processes based on current data collected from the Regional Healthcare System, and we studied the behavior of the processes when dealing with hypothetical scenarios. The latter ones have been defined according to the indications of experts, who were interested in gathering insights regarding both the quality of the interactions between process resources and the performance in case of increased amount of patients in severe conditions.

We exploited the simulation functionalities provided by the Signavio Process Editor [18], which allows one to directly open the process model in the simulation environment, without the need for further configuration. Decision logic has been simulated with the Camunda DMN Simulator [73], which allows one to verify the output of a decision table, given custom input values.

Signavio supports three kinds of simulation of syntactically sound BPMN processes (verification of the syntax is done by the tool prior to simulation), namely step-by-step, one-case, and multiple case simulation. Step-by-step simulation is guided by the modeler, who chooses which process flows are enabled, and it is mainly used for verifying that the process captures all the possible desired execution behaviors. This is fine for representing and observing the process flow and, therefore, it can be used when interacting with clinicians for improving common understanding. One-case simulation allows one to interactively reproduce the execution of a single process instance, according to specific conditions defined on process execution. Conversely, multiple-case simulation supports the run of multiple instances of the same process model, thus allowing one to aggregate results and analyze the overall workload of the process. In all circumstances, Signavio allows flow-related tokens to be graphically visualized, so that partial results can be visually monitored by process modelers and stakeholders.

The simulation tool requires the following information in input:

- 1. The probability associated to each outcome of an exclusive gateway (e.g., the proportion of COPD patients in stage 3 for gateway stage 3? in Fig. 6);
- 2. The fixed costs associated with each activity (e.g., spirometry performed in hospital costs around 25€);
- 3. The probability distribution that describes the duration for each activity (e.g., arterial blood gas analysis lasts 10 min, on average);
- 4. The hourly cost of each resource and the corresponding working schedule.

Data regarding the costs and duration of examinations have been collected by regional hospitals, by analyzing scheduled events and medical reports. On the contrary, the number of primary and hospital care resources involved in the management of COPD is managed directly by the regional healthcare system. According to regional data and experts' knowledge, 185,000 patients are diagnosed every year in the Veneto region. To obtain a reliable estimate of the amount of time that each resource devotes to COPD care, we considered that COPD patients in Veneto correspond to the 7,9% of the overall regional population. Accordingly, we assumed that the same proportion of patients is assigned to each practitioner, as in Italy each general doctor is responsible for the same amount of patients. Of course, we had to take territorial variance into account, as the percentage of ill people increases up to 11,8% in polluted areas. The same strategy has been adopted for estimating the time that pulmonary specialists working in hospitals dedicate to COPD patients. Thus, we considered having 3000 pulmonary specialists, working 15,000 hours per week with COPD patients, and 3500 general practitioners, working 58,000 hours per week with COPD patients (i.e., on average 1 h and 20 min of work for each general practitioners).

We created multiple simulation scenarios, starting from those that reproduce current reality to interesting what-if scenarios, suggested by experts. More particularly, we discuss here some simulations related to processes *COPD Diagnosis and Assessment* and *Management of COPD Exacerbations*, considering different aspects as (i) the proportion of spirometry test executed in PCUs with respect to those executed in hospitals and (ii) the increase in the number of patients diagnosed with COPD during winter, for process *COPD Diagnosis and Assessment*, while focusing on (iii) longer hospitalization following exacerbations episodes for process *Management of COPD Exacerbations*.

#### 4.5.1 Simulating Process COPD Diagnosis and Assessment

The first multiple case simulation focuses on the overall yearly cost of the process, depending whether spirometry tests are executed either in hospitals or in PCUs. To perform such simulation, we changed the proportion of patients following different paths according to gateway Spirometry prescription or direct request? in Fig. 6, which specifies whether the spirometry has to be done at the hospital or by PCUs. In Fig. 13, the difference between the costs associated to the performance of spirometry are shown. As discussed in [10], spirometry should be preferably performed in PCUs, as the costs of hospital spirometers is higher and waiting lists should be taken into account. Indeed, spirometry tests are done with more expensive devices, that should be reserved for severe respiratory complications.

A second simulation deals with the non-homogeneous distribution of COPD diagnoses during the year. Indeed, conditions of patients tend to worsen due to the cold weather and evidence suggests that in Veneto there are 15% more cases of COPD diagnosed in winter than on average during the year. According to these observations coming from regional data and experts' knowledge, physicians and healthcare stakeholders were interested in having an estimate of resources and costs during such season. Thus, we simulated the process *COPD diagnosis and assessment* for 120 days, corresponding to the average period of cold weather, considering that 68,800 patients are diagnosed in the given period. Such amount of patients is derived by considering that there are five working days per week (i.e., 86 days in 120 day frame), and 800 patients are managed every working day, thus considering an increase of about 15% with respect to the average number of daily patients.



Fig. 13 Costs of the process *COPD diagnosis and assessment* based on changing proportions of spirometry examinations carried out in PCUs or in hospital



**Fig. 14** Simulation results shown in Signavio for scenario Winter, run on 68,800 instances over 120 days. The tool highlights a bottleneck corresponding to the general practitioner, who is responsible for diagnosing COPD

The Signavio simulation interface, shown in Fig. 14, suggests a bottleneck in correspondence of the general practitioner, who is overloaded with the first diagnostic assessment. Figure 14 depicts in the blue bar at the right of activity Case history and working diagnosis the value of 67,300 (lower than the total number of patients—68,800), which is the number of patients who can be managed in the given period according to the actual number of available physicians. To complete the results of simulation and to allow the comparison of different scenarios, the part on the right of Fig. 14 reports quantitative results from the simulation of the winter scenario and from the simulation of the standard one. In particular, the report depicts costs, total cycle time, which is the overall amount of durations needed to complete all the process instances, and resource consumption, that is the amount of time resources need to execute the process. All these results can be expanded and visualized in detail. For instance, when clicking on the highlighted red box related to Bottlenecks in Fig. 14, we are able to see that the workload for each general practitioner is estimated to be around 184 days full-time, which is an excessive amount compared to the considered working schedule for 120 days.

#### 4.5.2 Simulating Process Management of COPD Exacerbations

Simulation results of process *Management of COPD Exacerbations*, depicted in Fig. 8, obtained for different lengths of hospitalization, are outlined in Fig. 15. Costs shown in the histogram are calculated by summing the fixed costs of hospitalization with the costs associated to the resources. In detail, we simulated the case of having 72,000 patients experiencing COPD exacerbation and considered that the 34% of them is hospitalized. Moreover, we evaluated different average



Costs of patient hospitalization in case of COPD exacerbations

Fig. 15 Simulation results of process *Management of COPD exacerbations* considering different hospitalization lengths, run on 72,000 instances over 100 days, with 23,800 patients hospitalized

hospitalization lengths, ranging from 3 to 20 days. As expected, the costs of the process increases drastically when patients are hospitalized for a longer time. The distribution of the total cost of process *Management of COPD exacerbations* considering the most relevant tasks for expense calculation is shown in the pie chart of Fig. 16. The depicted results correspond to the execution of the process by setting an average hospitalization length of 15 days.

# **5** Related Work

In this section, we discuss related work by focusing on different approaches for healthcare process design. We begin with presenting work proposed by the community of BPM and information systems. Then, we focus on research efforts conducted in the field of medical informatics, oriented to the development of CIGs systems. Finally, we discuss and compare our contribution with the introduced approaches.



Fig. 16 Distribution of the total cost of executing process *Management of COPD exacerbations*, considering a hospitalization length of 15 days

#### 5.1 Healthcare Process Modeling in BPM

Process management approaches have been employed in healthcare to support the modeling and enactment of organizational interdisciplinary processes [6]. However, applications of process modeling approaches in healthcare are still modest, despite the suitability of BPM techniques for representing complex and multidisciplinary organizational procedures. In addition, existing approaches tend to target specific clinical domains rather than proposing general methodologies, as exhaustively surveyed in [3].

Healthcare is considered a challenging domain for BPM approaches, as flexibility is often required and processes heavily depend on both information and knowledge [1, 6]. According to [36], a knowledge-intensive process is a workflow whose conduct and execution are heavily dependent on knowledge workers performing various interconnected and knowledge-intensive decision-making tasks, whose management and execution requires specific support. The role of human knowledge is considered central in such processes and, due to the variability in their structure, they cannot be directly mapped to business process diagrams.

However, *organizational* aspects of knowledge-intensive processes can be captured by BPM techniques [3, 6], whereas *clinical* or *medical treatment* processes can suitably handled by case management or artifact-based approaches [25, 74–76]. For instance, the ADEPT framework [75] allows one to revise running process instances and schemata in order to support the management of clinical guidelines. Among the interesting features of ADEPT, modularity, ease of use, and management of temporal aspects stand out. In the context of case management, a clinical pathway for living donor liver transplantation was firstly modeled using BPMN in [76]. Then, in order to facilitate transferability of the modeled pathway to other hospitals and to include flexibility caused by differences in treatments due to local policies, the processes have been modeled by using the OMG standard Case Management Model and Notation (CMMN)[25]. This standard can be used to model specific cases, together with the data that drive their execution, in order to simulate the decisions taken by a knowledge worker during process run-time.

However, traditional BPM approaches are activity-centric, that is, they focus on representing the process control flow, by supporting structured and repetitive organizational processes. As for BPMN-based approaches, BPMN is used for modeling surgical pathology processes in [7]. According to the authors, the notation was chosen because it is widely accepted and recognized in industry, due to the easiness provided for the design of simple or high level processes. Indeed, they advocate that the obtained models have be as simple, transparent, and understandable by care providers.

Most research proposals focus on extending the core of the BPMN notation to improve its expressiveness with respect to the design of healthcare processes, often introducing new elements that are tailored for modeling specific clinical circumstances. In [77], the authors propose a conservative extension of BPMN for the healthcare domain, with the aim of incorporating role information and facilitate the assignment of tasks to performing resources. A major extension to BPMN, proposed to systematically support the design of clinical pathways, is developed in [78].

This extension, named BPMN4CP, incorporates concepts retrieved from the context of clinical pathways, with the aim of enhancing evidence-based decision activities related to the considered domain. New tasks types are added to model medical diagnosis and therapies, while different data objects types are used for representing documents related to both medical and administrative procedures.

Conceptual modeling of care pathways is addressed under different viewpoints in [79] and [80]. Specifically, in [79], a novel methodology for connecting processes and data belonging to a healthcare information system is proposed, based on the concept of *task view*, which represents the information that can be accessed by a process task and that is relevant for its execution. In [80], a data-aware representation of BPMN processes is presented as a means to support clinicians working in ICU to understand their responsibilities in the care process, how therapies are temporally constrained and which are the critical decisions that affect the quality of the provided care.

The processes designed in [80] have been further refined and used in [81, 82] as a blueprint for guiding the development of a rule-based decision-support system for antibiotic stewardship.

Finally, in [83], two clinical pathways for colon and rectum carcinoma are designed as part of a pilot project, with the aim of evaluating the benefits of using BPMN for modeling structured medical procedures.

A general methodology for the design of medical processes was recently presented in [84]. The authors present an approach based on three main phases for healthcare process modeling, starting from context analysis and addressing the design of conceptual and logical process models. UML activity diagrams are chosen to represent processes, in order to ease integration with other UML diagrams, such as class diagrams, suitable for representing other aspects of the clinical domain. Despite the motivations described in [84] are similar to those addressed in this paper, the two methodologies have different focuses. In [84], the integration of medical processes and health information systems plays a central role, whereas the modeling and management of decisions is not specifically addresses. Moreover, authors provide a mapping toward logical formalisms, while we focused mostly on conceptual design, yet completing it with organizational analysis based on KPIs and process simulation.

In general, all the presented approaches are directed to expert modelers and aim at improving understanding and, in some cases, standardization of care procedures under an organizational standpoint.

However, as processes are becoming more and more decision-intensive, the need for a more appropriate language to support the modeling of process-related decision has recently been recognized within the BPM community. In healthcare, user decisions and unpredictable events contribute to the continuous generation of new clinical knowledge, thus making the structure of the processes less rigid. The previously introduced BPMN-based approaches, do not explicitly deal with the design of process-related decisions, as these either tend to remain hidden within other process elements or are managed by dedicated decision support systems. In general, clinical decision support is often directed to medical personnel, whereas decision management is useful to serve organizational purposes.

DMN has been proposed to model different aspects of decision-making and to complement BPMN to this end. The recently proposed notation allows one to model

design in the context of collaborative networks in [41]. An approach to derive DMN decision models from BPMN process models is discussed in [85]. Upon having identified important decision steps in the process, decision logic is extracted from process logs, in the form of decision tables, and a compound DMN model is constructed with the aim of re-adapting the process diagram to fit the modeled decision logic. A logic-based formalization of DMN decision tables has recently been provided in [86], in order to unambiguously define a semantics for such constructs and introduce several analysis tasks, focused on correctness checking. However, at the best of our knowledge, DMN has not yet been applied for modeling clinical decisions, despite the suitability of decision requirement diagrams for the design and sharing of human decision-making. Indeed, decisions in healthcare are complex, shared among different professionals, and often rely on medical evidence and expertise. Despite the formalization of these aspects is quite complex, explanatory decision requirements diagrams can be used for compliance, standardization, and training and educational purposes.

Finally, business process re-engineering (BPR) requires a clear understanding and a radical re-design of organizational procedures, in order to overcome critical steps and to improve process performance, flexibility and adaptability to change. Most reengineering approaches focus on re-thinking the process control flow, in order to allow parallel execution of activities and resource re-allocation with the aim of reducing execution times and costs. In this scenario, data and decisions are often managed and re-engineered independently and only information related to process metrics is employed for this scope. Examples of canonical applications of process and knowledge re-engineering in healthcare are discussed in [8, 87, 88]. In detail, in [8] a framework based on event driven process chains, the entity-relationship model, and discrete event simulation is introduced in order to define and analyze the current state of a surgical ward and design an improved system. A more general study addressing the identification of those business process re-engineering principles that can improve the application of E-Health technologies within a healthcare environment is conducted in [87].

# **5.2** Clinical and Healthcare Process Modeling: Clinical Interpretable Guideline Approaches

Despite the complexity of medical and clinical processes, the community of medical informatics has been actively investigating and developing systems for the modeling and management of clinical practice guidelines [3].

Computer-interpretable guidelines (CIGs) are formalizations of clinical practice guidelines aimed at fostering the development of guideline-based decision support systems [89]. In general, the implementation of guideline-based support systems requires timely data sharing, to ensure communication between care providers, data completeness, and correct formalization of guideline rules and statements [90]. In

addition, proper clinical knowledge and decision management features must be provided for dealing with knowledge-intensive care processes.

According to [91], the following areas of development must be considered when modeling clinical guidelines: (i) modeling and representation, (ii) acquisition, (iii) verification and testing, and (iv) execution and monitoring. The most relevant CIGs formalisms are exhaustively reviewed in [89]. Here, we focus on discussing workflow-based approaches and task-network models, as the structure of the latter ones can be compared to the concept of process control flow.

Workflow and process modeling techniques have been applied for specifying, designing, and implementing procedural aspects of clinical guidelines. One of the most complete projects addressing the integration of clinical guidelines into organizational workflows is the GUIDE project, developed within the Careflow framework [92]. GUIDE is a multi-level architecture designed to integrate a formalized model of the medical knowledge retrieved from clinical guidelines with workflow management systems. The presented approach is based on computational formalisms that facilitate the work of healthcare stakeholders by suggesting them the medical task to execute and the resources to allocate.

In [51], the authors introduce a temporal model for fostering the conceptual design of clinical workflows for stroke prevention and management, by addressing activity duration, delays, periodic and absolute constraints, and inter-activity constraints.

As for CIGs formalisms, several languages for clinical guideline modeling have been proposed, such as GLIF [93], Asbru [94], PROforma [95], and SAGE [96].

The guideline interchange format (GLIF) was proposed to ease guideline sharing [93]. The core language includes several ontological entities that allow the representation of guidelines as multi-step flowcharts that support branching logic. The current core, GLIF3, supports the modeling of guidelines at different abstraction levels, namely conceptual, computable, and implementable. An interesting aspect of GLIF3 is the hierarchical modeling of decisions, which distinguishes between automated decision steps and those that are made by a health worker. Asbru [94] enables designers to represent clinical guidelines, by considering continuous prescribed actions, and by supporting the parallel, in sequence, or ordered execution of care plans. Sub-plans and their outcomes can also be specified. PROforma [95] allows one to represent guidelines as directed graphs of tasks, which can be either plans, decisions, actions, and enquiries. Scheduling and temporal constraints define task ordering, and task execution is based on the triggering of pre-/post-conditions. Finally, SAGE [96] builds upon previously introduced work [93-95] and incorporates workflow representation, information, and terminology standards, and control flow standards. Likewise GUIDE [92], it fosters the definition of an organizational model, with resources and roles. Such model and the guideline model are kept separate, to ease the adaptation of the guideline to local organizational realities.

A recent architectural framework for promoting the design, implementation, and evaluation of continuous guideline-based decision support is presented in [97]. A decision support engine, called PICARD, interacts with a guideline knowledge library and a temporal reasoning service in order to apply medical knowledge to patient data during the execution of clinical tasks. This framework provides alerts and recommendations at the point of care to health workers, patients, and

knowledge engineers. In detail, patients treated in home-care settings can connect via a mobile client application to obtain personalized recommendations. Care providers such as physicians or nurses receive alerts, or therapy recommendations based on telemedicine data. Knowledge engineers can perform simulations on different guideline-related scenarios. In addition, the architecture allows accesses for the retrieval of procedural knowledge, usually expressed in terms of workflows. Currently, PICARD is the backbone for MobiGuide [98], an intelligent decision-support system for patients affected by chronic diseases. Briefly, patients are provided with sensors, which collect real-time information that is integrated with clinical history data and medical knowledge in order for supporting better decision-making. Alerts and personalized guideline-based recommendations are sent to patients, through their mobile phones.

A further contribution from the MobiGuide project is described in [99], where the authors consider different strategies for integrating CIGs with business processes (organizational workflows, in the authors' terminology) and propose the implementation of some use cases in the domain of atrial fibrillation. Such implementation is based on a loosely-coupled integration of CIG and BP engines, mediated through the health record database system. With respect to our proposal, which focuses on methodological issues, in [99] the authors pay attention to more implementation-related issues, with the use of several healthcare-related standards as IHE, HL7, and other XML-based general standards. From this point of view, no attention is paid to consider the expressivity of BPMN advanced constructs, as events, exceptions, complex gateways, and so on with respect to the specification of care pathways. Moreover, no methodological issues are discussed with regard to the design of care pathways. As for more implementation-oriented issues, standards such as HL7 [100], ICD9-CM [101], or SNOMED CT [102] can be used to encode knowledge in DMN requirement diagrams. In addition, they could be used as values for FEEL expressions, as FEEL supports several kinds of data structures and its semantics draws inspiration from Java, JavaScript, XPath, SQL, PMML, Lisp, and many others [21]. In particular, FEEL is thought to be toolindependent, executable, and expressive enough to support all kinds of decision logic.

Among the first studies that considered both medical records, workflows, and HL7 standard to deal with the derivation of quality indicators, we mention here [103], where authors underline that workflow aspects must be considered and modeled to evaluate quality indicators. Such kind of results may be considered as a motivation of our methodological proposal as it confirms that process-related aspects need to be considered to gether with clinical data.

Generally speaking, despite both CIGs and BPM approaches aim to improve healthcare processes and decision-making, a few crucial differences can be discerned. CIGs are meant to be used by clinicians and, thus, they focus on clinical decision support and on the development of solutions tailored for dealing with a specific clinical guideline, enacted in a well-defined context. On the contrary, BPM approaches in healthcare aim to foster standard-based tools for supporting broader organizational as well clinical aspects in healthcare, handled by physicians, healthcare professionals, and IT and organization experts. In this paper, we provide an agile design methodology, based on the well-known OMG standards BPMN and DMN, that addresses the integrated yet complimentary design, implementation, and enactment of care processes and related decisions, considering the viewpoint of an expert modeler. The introduced step-wise approach promotes the iterative and agile design of organizational aspects of care pathways, mainly by focusing on the standardization of processes and decisions. The methodology supports incremental refinements, as the modeler can choose how and how many steps to apply, depending on the final objectives. Procedural, decisional, and informational aspects of care pathways are represented in an integrated and scalable format, thus supporting the re-use of clinical data and knowledge for process and decision re-engineering. To our knowledge, this is the first time a structured approach based on the joint use of BPMN and DMN has been applied to the healthcare domain.

Compared to the discussed contributions belonging to the BPM community, the methodological approach introduced in this paper utilizes conceptual modeling techniques for purposes that go beyond clinical process documentation and information sharing. Indeed, simulation and indicator-based process analysis are discussed, and the role of information in process management is promoted to first citizen in the methodology. The integration of standard process modeling techniques with systematic decision and information design, possibly used for re-engineering purposes, is a novelty of the introduced methodological framework. In addition, the methodology is presented at a quite high level, thus allowing the integration of other approaches, suitable to meet specific requirements. For instance, having used the OMG notations BPMN and DMN, facilitates the integration of CMMN models, in case there is the need to deal with more flexible clinical aspects. Similarly, standard ontologies and terminologies can be integrated in the BPMN and DMN processes, for defining domain-specific concepts.

Whereas traditional process management techniques are suitable for representing *organizational* aspects and procedural process knowledge, computer-interpretable guidelines approaches aim to support and facilitate appropriate decision-making at the point of care. Usually, the main goal of such systems is to provide direct decision support to clinicians, by fostering clinical knowledge representation. For this reason, CIGs are suitable to capture complex knowledge-based aspects of *clinical* and medical treatment processes. However, the main drawback of CIGs approaches is their lack of standardization [3, 89], which motivated the introduction of workflow-based approaches for better modeling organizational aspects [92, 96]. Indeed, despite the abundance of language proposals, no specific proposal emerges over the others, and each approach focuses on a selected set of modeling aspects.

The goal of the proposed methodology is to start from a standard, modular, and extensible core, that allows one to exploit existing standard tools for healthcare process modeling. The benefits of using standard tools are manifold. First of all, process and decision modeling are based on graphical and understandable notations, that can be conservatively extended and integrated with other (standard) approaches to suit the need for flexibility required in the healthcare domain. For instance, new types of tasks can be created in conformance to the standard, as done in [78], and process

cases can be added to handle run-time variability [76]. Secondly, process exchange, deployment, and verification are supported by several tools, which may already be used by the administrative counterpart of healthcare organizations. Besides, modelers are facilitated by the existence of numerous control-flow, data, temporal, and resource design patterns [38, 39, 55, 104].

In general, BPMN is a powerful language for capturing various procedural aspects, as it provides a wide range of constructs suitable for advanced event and exception handling. Especially in the context of temporal constraints, researchers are actively proposing approaches for the verification of a process temporal properties and for the resolution of related violations. Motivated by processes taken from the domain of emergency medicine, in [49] the authors propose an approach for representing and verifying temporal constraints. These are expressed by extending a basic BPM modeling notation and suitable algorithms have been proposed to verify dynamic properties of temporally-enhanced business processes. TNest is a structured and modular workflow modeling language providing full support of temporal constraint specification during process design, presented in [9]. In particular, TNest allows one to define two main kinds of temporal constraints, that is, activity duration and relative constraints. Finally, in [53] temporal constraints in the context of modularized processes are investigated. In this setting, the re-use of process knowledge must be enabled and, thus, authors consider the issue of representing the overall temporal behavior of a sub-process. This way, temporally enhanced sub-processes can be used within other processes, hiding the internal temporal features of such subprocesses.

Yet, born to capture process control-flow, BPMN does not fully support other perspectives, such as the representation of domain knowledge and the integration of complex, structured data [6, 105]. Despite research efforts are devoted in this direction considering clinical domains [11, 56, 75, 76], the expressiveness of CIGs remains more suitable for dealing with complex clinical aspects, especially when developing tools that require clinical knowledge and data representation and the direct involvement of physicians and patients [98].

# 6 Conclusions

In this paper, we introduced a methodological framework for supporting the design, implementation, and enactment of decision-intensive healthcare process, by exploiting advanced process design techniques and by integrating them with proper decision modeling. We advocate that combining the advantages derived from the design of process and decision models, each one stressing on a different care perspective, increases the overall value that such a synergistic design approach can bring to care pathways development. In addition, the proposed methodology provides a preliminary classification of the kinds of data involved during the different phases of process development. Proper information and knowledge management allows stakeholders to better support iterative process design, execution, and re-engineering, thus leading to more flexible and information-aware care pathways.

The advantages brought by the proposed methodology for care pathways design have been compared to those achieved by computer-interpretable guidelines for clinical process modeling. We concluded that, whereas CIGs are suitable to build for decision support systems directed to clinicians, the use of BPM approaches fosters the definition of a standard-based core of care pathways and related information, able to address crucial aspects of healthcare organizations. Given the technological and market-related scenario discussed for BPM in the Introduction, we could easily compare the current situation for BPM systems and healthcare to the situation of database management systems (DBMSs) and medicine in the late 80s and in the early 90s. At that times, many research efforts were related to how to use and extend such database methods and tools to manage (even complex) clinical data. Now, it is completely obvious that any software system managing large amounts of clinical data must rely on the design and realization of a suitable database management of care pathways will rely and will have to consider BPM methodologies and technologies. According to this observation, our proposal may be considered as a first attempt to highlight specific issues we have to consider when designing decision-intensive care pathways.

For future work, we intend to further investigate the relationships between processes and healthcare data models, in order be able to detail relevant data at a sufficient level of abstraction, suitable for implementation and integration within existing healthcare systems.

**Acknowledgements** The authors would like to thank Enzo Rizzato MD, Manager of the 4<sup>th</sup> Local Health and Social Care Facility of Veneto and Maria Cristina Ghiotto MD, in charge of the Territorial Assistance and Primary Care Units for the region of Veneto for involving us in the development of COPD care pathways. We would also like to thank our student Sonia Frei for contributing in the analysis of the designed COPD processes.

#### Appendix A: Modeling COPD Subprocesses

#### A.1 COPD Diagnosis and Assessment

Diagnosis of COPD has to take into account a multiplicity of signs and symptoms. A clinical diagnosis of COPD should be considered in any patient who has dyspnea, chronic cough or sputum production, and presents a history of exposure to risk factors for the disease [23]. Among genetic disorders that must be considered in a diagnosis of COPD, the best documented is  $\alpha_1$ -antitrypsin deficiency, which leads to an increased predisposition to obstructive pulmonary disease [106].

Spirometry is required by international clinical guidelines to make a diagnosis of COPD [23, 57]. Practically, a spirometer measures how quickly full lungs can be emptied and the total volume of air expired. Spirometric measurements used in a diagnosis of COPD include:

- Forced vital capacity (FVC), which is the maximum volume of air that can be exhaled during a forced maneuver.
- Forced expired volume in one second (FEV<sub>1</sub>), which is volume expired in the first second of maximal expiration, during a forced maneuver. This value measures how quickly lungs can be emptied.

FEV<sub>1</sub>/FVC, which is the ratio between the two previously calculated values.
 FEV<sub>1</sub> is expressed as a percentage of the FVC and gives an indication of airflow limitation, called Tiffenau index [23].

Spirometry results are evaluated by comparison with reference values calculated on healthy subjects and depending on age, height, gender, and ethnicity.

Currently, the spirometric criterion used for diagnosing COPD is a postbronchodilator FEV<sub>1</sub>/FVC ratio below 0.7. Bronchodilators are medications that increase the FEV<sub>1</sub> or change the other spirometric variables, by altering airway smooth muscle tone. Performing spirometry after the administration of a short-acting inhaled bronchodilator helps to differentiate between an acute or persistent obstructive defect and minimizes outcome variability [23]. However, the aforementioned threshold is age-dependent and can lead to a significant degree of overdiagnosis in the elderly and of underdiagnosis in younger patients. Besides, as multiple respiratory diseases present a FEV<sub>1</sub>/FVC ratio < 0.70, differential diagnoses must be considered in order to determine an appropriate treatment [107]. Chest X-rays may be prescribed to confirm a working diagnosis of COPD and questionnaires such as the COPD assessment test (CAT) are administered for assessing symptoms. Indeed, when used in conjunction with spirometry results and demographic data, these tools have proven to enhance diagnostic accuracy [59].

For grading COPD severity, spirometric classification remains the preferred standard method [23]. Post-bronchodilator lung function, expressed in terms of  $FEV_1$ and  $FEV_1/FVC$  ratio, is used to classify COPD patients into four levels of severity [23], summarized in Table 2. Patients showing a post-bronchodilator Tiffenau index greater than 0.70 are not considered in COPD management, despite those who smoke or have exposure to pollutants are considered at risk and may be targeted by prevention programs.

The BPMN process corresponding to *COPD diagnosis and assessment*, depicted in Fig. 6, consists of a main pool, which stands for the regional healthcare system. This is partitioned into two lanes, one representing a primary care unit and, specifically, a general practitioner, the other one representing a pulmonologist working in a regional hospital.

Since one of the goals of the COPD process models is to define the respective responsibilities of general practitioners and pulmonary specialists, only these two roles have been represented explicitly. Indeed, practitioners and pulmonologists have to supervise the performance of nurses, technicians, and auxiliary personnel who collaborate in the process.

Table 2         Levels of COPD           severity, based on spirometric         assessment, taken from [23]	Severity	FEV <sub>1</sub> /FVC	FEV <sub>1</sub>
	Stage 1 (Mild)	< 0.70	$\geq 80\%$
	Stage 2 (Moderate)	< 0.70	[50%, 80%)
	Stage 3 (Severe)	< 0.70	[30%, 50%)
	Stage 4 (Very severe)	< 0.70	< 30%

.. . .

207

The process begins when the patient consults the general practitioner, typically after having experienced some sort of respiratory discomfort, such as productive cough, acute chest pain or breathlessness. Case history and working diagnosis is the first decision activity encountered, modeled as a user task, which represents the gathering of useful observations during medical inspection that are analyzed in conjunction with the patient clinical history and exposure to risk factors in order to formulate a working diagnosis of COPD.

All the collected information regarding symptoms, smoking habits, and risk factors is recorded in the local PCU database.

The following exclusive gateway, COPD Suspected? directs the flow toward the prescription of spirometry and further examinations, if a diagnosis of COPD is considered, or toward the process end event, otherwise. Spirometry (in PCU)/Spirometry (in hospital) is indicated to support COPD diagnosis, to assess the severity of airflow obstruction and, in some cases, it can be used as a prognostic indicator. If patients in primary care have direct access to spirometers, a simple spirometry test is requested and performed by a general practitioner directly within a primary care facility. Otherwise, patients need a prescription for the spirometry to be performed in hospital.

Under an organization standpoint, the biggest difference between the performance of a spirometry in PCU or in hospital is waiting time. Whereas in PCU, spirometric assessment can be executed during a routine examination, spirometers in hospitals are used for assessing various respiratory problems and patients are prioritized according to the severity of their conditions.

Chest X-rays may also be carried out in hospital, if believed necessary to support the diagnostic process. Spirometry allows measuring FEV<sub>1</sub>, FVC, and the ratio between such values. The physician must read and record spirometry results received with the corresponding reports, and compare them with reference values for age, weight, height, gender, and ethnicity. If airflow obstruction is detected, the practitioner must prescribe post-bronchodilator spirometry (PBS) to confirm the diagnosis of COPD and to measure the degree of airflow obstruction and its reversibility. Once the patient has been informed and advised about the conduct of the test, postbronchodilator spirometry is performed by a pulmonologist in hospital. Subsequently, the pulmonologist must write PBS report, containing the diagnostic statement and the spirometric results. This is sent to the general practitioner, who must confirm the diagnosis of COPD and record the patient data in the primary care unit database, following the ICD-9-CM coding rules [108]. The general practitioner can also prescribe pharmacological treatment, whenever considered necessary to relieve the patient from symptoms. Then, the COPD assessment test (CAT) is administered to gather information about the impact of COPD on the patient's quality of life and to measure general health status impairment. If the patient is a smoker, additional tests, i.e., Mondor and Fagerstörm, are administered to reveal addiction to tobacco consumption and nicotine [109].

Once the test-based assessment is concluded, the physician needs to Inform the patient about his/her clinical condition and can suggest a detoxing treatment in specialized facilities for smoking cessation. Then, the patient undergoes a pulmonary examination in hospital, which results in COPD staging, based on the outcomes of

the previously executed spirometric assessment. The staging report is authored by the pulmonologist and must contain the assessed COPD stage, the value of the Tiffenau index, an indication for medical examinations and future re-evaluations, and therapeutic advice. The general practitioner receiving the report must verify that the assessed stage is correct before proceeding with COPD stage recording as he/she is responsible for signaling possible incongruities. Indeed, the classification of COPD patients is a crucial decision as the planning of the future inspections, both in terms of the kind of medical analyses required and their frequency, depends on the severity of the assessed grade. If the patient is in stage 3, a pulmonary specialist in hospital is put in charge of monitoring the patient. In this case, the pulmonologist must prescribe pharmacological therapy, schedule pulmonary examination, schedule post-bronchodilator spirometry, and schedule 6-min walk test (6MWT) to monitor the disease, and to evaluate the patient's functional exercise capacity. Conversely, if the patient is assessed with stage 1 or stage 2, PCUs remain in charge of COPD management. PCUs must also provide a counseling service in order to educate patients to avoid exposure to behaviors that can further compromise their health condition. The progression of the disease can be limited by efficient monitoring and changes in lifestyle and smoking habits. In the final stage of COPD assessment, the general practitioner needs to schedule dietitian consultation if the patient presents a BMI greater than 30. Finally, he/she has to plan vaccinations (influenza and pneumococcus), as they appear to be effective in older and severe patients [23]. Once the main steps of the care intervention have been scheduled, the COPD diagnosis and assessment process terminates.

#### A.2 Ordinary Management of Stable COPD

Life-long and continuous management of stable COPD can become quite complex especially in advanced stages of the disease, as multiple complications have to be considered beside compromised pulmonary function. In particular, patients affected by COPD suffer from impaired gas exchange and are subjected to develop cardiovascular, metabolic, and neoplastic comorbidities [23].

Therefore, COPD treatment aims to prevent the progression of the disease, relieve patient symptoms, improve exercise tolerance and general health status, and prevent and treat correlated complications. The type of care workers in charge of COPD management and the frequency of medical inspections depend on the severity of the disease, on the patient individual response to pharmacological treatment, and on the healthcare system. Ordinary management of COPD encompasses the following important aspects: smoking cessation, optimization of pulmonary status by pharmacological therapy, improvement of exercise tolerance, nutritional care, and possible support in terms of oxygen therapy or pulmonary ventilation. Besides, education, social, and behavioral aspects must be considered to improve compliance [57]. Vaccinations can be required, based on local policies, availability, and affordability [23]. Influenza and pneumococcal vaccinations are planned yearly, to prevent the insurgence of serious illnesses that may contribute to COPD exacerbations.

Ordinary management of COPD is based on individualized assessment and aims to reduce both current symptoms and risk of sudden worsening [23]. As previously

mentioned, the BPMN diagram representing the main steps of ordinary management of COPD, depicted in Fig. 7, is executed with a periodicity that depends on assessed COPD stage. This periodicity is captured by a start timer event. Specifically, the process is expected to start every 2 years for patients in stage 1, every year for patients in stage 2, and every 6 months or less for patients in stage 3.

The first steps of the care process are executed either by the general practitioner, for patients in stage 1 and 2, or by the pulmonologist, for patients in stage 3, but the conduct of the explained actions coincides. Firstly, it is necessary to verify administered vaccinations. In the studied context, vaccinations are managed by district health authorities. If vaccinations have not been administered, the physician must reschedule vaccinations. If the patient refuses to be vaccinated, the motivation behind the refusal must be recorded in the health authority information system HC district db. In the studied context, data suggest that elderly patients often forget to attend vaccination appointments or they have troubles reaching care facilities, due to mobility limitations. In this latter case, vaccinations can be provided at home, depending on local service availability.

To monitor the disease, care providers are required to administer COPD assessment test (CAT) and to adjust current bronchodilator therapy, if needed. Pharmacological therapy is prescribed according to the severity of the illness, the availability of the drugs, and the presence of comorbidities. Treatment regimen should be patientspecific as the relationship between the severity of symptoms and airflow obstruction tends to vary from patient to patient. Moreover, COPD severity is also dependent on the geographical distribution of the population. Usually, patients in stage 1 are treated with short-acting inhaled bronchodilators, while long-acting bronchodilators are prescribed to patients in stage 2 or 3, in different dosage and combinations.

Then, different inspections are carried out in parallel, to evaluate the patient's health status, and re-assess BMI and respiratory function. Mondor and Fagerstörm tests are administered in order to evaluate lifestyle and smoking habits, with the aim to assess the patient's progress toward smoking cessation. Physicians must also evaluate dyspnea and exercise tolerance. The objective measurement of exercise impairment is obtained with the help of the 6-min walk test [24]. This is used in conjunction with the Modified British Medical Research Council (mMRC) dyspnea questionnaire to quantify the degree of disability due to breathlessness during daily activities. Body mass index (BMI) must also be re-assessed, in order to decide if it is necessary to Prescribe a dietitian consultation. Obesity and weight loss, when combined with muscle wasting, contribute significantly to morbidity, disability, and handicap in COPD patients, who must receive nutritional therapy. Pulse oximetry is used to evaluate oxygen saturation. If the measured oxygen saturation is less than 92%, blood gas assessment is scheduled. Arterial blood gas analysis (ABG) is the preferred method for determining the need of oxygen therapy [57]. Finally, the patient is examined and all analyses results are resumed and recorded in the dedicated information system PCU db.

If the patient is assessed with stage 2 or stage 3, he/she might necessitate an additional pulmonary inspection. In this case, a pulmonologist can request oxygen therapy according to the considered clinical picture. Long-term oxygen therapy is usually administered continuously and in a home-care setting to improve survival,

exercise, sleep, and cognitive performance. In addition, a cardiological consult may be requested, to exclude cardiovascular instability. Finally, the pulmonologist must review COPD stage in order to confirm the stage after treatment or to update it. At the end of each instance of the COPD ordinary management process, the general practitioner must either confirm COPD stage or update COPD stage, if this has changed. In case of stage change, signal event re-schedule is thrown to be caught by the corresponding interrupting boundary event, depicted in Fig. 5 and attached to the border of sub-process management of COPD exacerbations.

The introduced healthcare process representing the ordinary management of stable COPD is periodically re-enacted and the patient is also involved in educational programs to improve recognition of symptoms worsening and self-care.

#### A.3 Management of COPD Exacerbations

COPD exacerbations are "acute episodes characterized by a sudden worsening of the patient respiratory symptoms that is beyond expected day-to-day variations" [110]. The main drawback associated with COPD exacerbations is represented by hospital admissions, that negatively impact the disease evolution, economical costs, and quality of life [67].

Several clinical findings must be considered when evaluating patients with exacerbations. Among these COPD severity, the presence of comorbidities, and the history of previous exacerbations are the most important. Indeed, patients that are subjected to two or more exacerbations of COPD per year seem to maintain this worsening trend over time and, thus, must be constantly monitored. A diagnosis of exacerbation requires a physical examination to evaluate the effects of the episode on both cardiovascular and respiratory systems [23].

COPD exacerbations are often treated with temporary short-acting inhaled beta2agonists, which can be combined with corticosteroids and antibiotics. However, if the patient does not respond to outpatient pharmacological treatment, hospitalization is required. When a patient is admitted to hospital, supplementary oxygen is administered, short-acting bronchodilators are prescribed, and pulmonary rehabilitation may also be indicated [57].

COPD hospital admissions due to exacerbations negatively affect the patient quality of life and the progression of the illness, while substantially increasing the costs of patient management [67]. However, a standardized and integrated care intervention, supported by proper data collection, and resource coordination, seems to effectively prevent hospitalizations due to exacerbations in COPD patients [67, 111].

The goal of exacerbation treatment is to minimize the effects of the on-going worsening and prevent future episodes. The process enacted for treating COPD exacerbations in Veneto is depicted in Fig. 8.

An exacerbation episode is usually detected by the general practitioner, following the clinical presentation of the patient, who complains of an acute change of symptoms. Diagnosis of exacerbation relies on the current symptoms, on the evidence gathered during the physical examination with pulse oximetry, and on the history of previous worsening episodes. If more than two exacerbations occur within 1 year, the stage of the disease is expected to vary, as the patient respiratory function is compromised more rapidly [57]. In general, diagnosing an exacerbation can become quite cumbersome, as multiple physical findings and diagnostic procedures must be evaluated, with the main aim of excluding the development of comorbidities. If oxygen saturation measured with pulse oximetry is < 90%, the exacerbation is likely to be life-threatening and, thus, hospital and intensive care treatment is indicated. Otherwise, the general practitioner can suspend current therapy in order to begin therapy to oppose worsening. Usually, short-acting beta agonists (SABA), short-acting muscarinic antagonists (SAMA), inhaled corticosteroids, and antibiotics are prescribed, according to the considered clinical picture. Then, after 24/48 h, the patient is reevaluated. Following clinical diagnosis, gateway Is exacerbation life-threatening? determines if the patient needs to be treated in inpatient or outpatient settings. If the patient has improved, the exacerbation can be managed in an outpatient setting. In these circumstances, it is indicated to continue with current therapy and to schedule periodic examinations to ensure proper patient monitoring. If the patient has not improved after treatment or oxygen saturation is assessed < 90%, he/she is sent to the hospital emergency room. Hospitalization must be considered when the patient cannot been treated adequately in home care settings, due to the severity of the respiratory dysfunction [23, 57]. Following ER triage, ER physicians must test hydration level to ensure proper fluid balance and they need to assess symptoms. Then, the patient is physically examined and continuously monitored. Physicians in ER must exclude other clinical complications, such as heart failure, bronchial pneumonia, pulmonary embolism, and pleural effusion. A pulmonologist examination in ICU can be requested to assess arterial blood gases and execute chest radiography. If the severity of respiratory dysfunction is high, the patient must be admitted to an intensive care unit and treated under the responsibility of a pulmonologist. Different factors must be considered when evaluating hospitalization. Typically, patients are admitted in ICU when COPD symptoms are acute, there is a preponderance of sensory disturbances, there is evidence of new cardiac arrhythmia, or no response to exacerbation treatment is observed [23]. The treatment to solve exacerbation is mostly based on oxygen therapy, which aims to prevent tissue hypoxaemia, i.e., reduced oxygen availability, and at preserving cellular oxygenation. During hospitalization, the general practitioner is informed about the administered care. This is necessary to provide practitioners with all the information, needed to optimally treat the patient after hospital discharge. Patient discharge is decided after evaluating several criteria, which consider also local policies and availability. The discharge report must contain information regarding the possibly changed COPD stage, respiratory function, comorbidities, outpatient suggested therapy, and clinical follow-up. Patients dismissed with hypoxemia or in stage 3 may require short-term or long-term oxygen therapy. In this case, a pulmonologist must take the patient in charge and a home-care oxygen therapy protocol is started. Oxygen therapy is monitored by specialized centers in the region and its efficacy must be re-evaluated yearly.

As required by the national program for the evaluation of care outcomes, 2 weeks after patient discharge, the general practitioner must verify treatment efficacy and effectiveness. The aforementioned evaluation program aims to estimate the overall quality of the provided care, in terms of treatment costs, appropriateness, and efficacy, in order to compare the outcomes of the overall care intervention with audit and bench-marking data. Hospital re-admission of COPD patients within 30 days from discharge is probably the most important indicator used for assessing the quality of the overall care plan [67, 112]. In the studied context, around 15% of discharged patients is re-hospitalized within 30 days from discharge. During this final verification phase, the stage of COPD is also reviewed and, if it has changed as a consequence of the exacerbation, signal state changed is thrown. This is caught by the corresponding non-interrupting boundary event attached to the border of sub-process management of COPD exacerbation of Fig. 5, which enables an exception flow devoted to the rescheduling ordinary examinations. Likewise ordinary management of COPD, exacerbations require follow-up. This includes reassessment within 4 weeks, evaluation of improvement in symptoms, physical examination, assessment of need for oxygen therapy, and re-adjustment of current treatment regimen.

# References

- Dadam P, Reichert M, Kuhn K (2000) Clinical workflows the killer application for processoriented information systems? In: BIS 2000: 4th International Conference on Business Information Systems, pp 36–59, Poznań, Poland. Springer, London
- Combi C, Pozzi G, Veltri P (eds) (2018) Process modeling and management for healthcare. CRC Press, Taylor & Francis Group, Boca Raton
- Russo A, Mecella M (2013) On the evolution of process-oriented approaches for healthcare workflows. Int J Bus Process Integr Manag 6(3):224–246
- 4. Schrijvers G, van Hoorn A, Huiskes N (2012) The care pathway: Concepts and theories: An introduction. International Journal of Integrated Care, 12(SPECIAL EDITION I). cited By 19
- Campbell H, Hotchkiss R, Bradshaw N, Porteous M (1998) Integrated care pathways. BMJ: British Medical Journal 316(7125):133
- Lenz R, Reichert M (2007) IT support for healthcare processes premises, challenges, perspectives. Data Knowl Eng 61(1):39–58
- García Rojo M, Rolón E, Calahorra L, Óscar García F, Sánchez RP, Ruiz F, Ballester N, Armenteros M, Rodríguez T, Espartero RM (2008) Implementation of the business process modelling notation (BPMN) in the modelling of anatomic pathology processes. Diagn Pathol 3(1):S22
- Bertolini M, Bevilacqua M, Ciarapica FE, Giacchetta G (2011) Business process re-engineering in healthcare management: a case study. Bus Process Manag J 17(1):42–66
- Combi C, Gambini M, Migliorini S, Posenato R (2014) Representing business processes through a temporal data-centric workflow modeling language: An application to the management of clinical pathways. IEEE Trans Syst Man Cybern Syst 44(9):1182–1203
- Combi C, Oliboni B, Zardini A, Zerbato F (2016) Seamless design of decision-intensive care pathways. In: IEEE International Conference on Healthcare Informatics (ICHI 2016), pp 35–45. IEEE
- Combi C, Oliboni B, Zerbato F (2017) Towards dynamic duration constraints for therapy and monitoring tasks. In: Proceedings of the 16th Conference on Artificial Intelligence in Medicine, AIME 2017, pp 223–233. Springer International Publishing, Vienna
- 12. Stratistics MRC (2017) Business Process Management (BPM)-Global Market Outlook (2016–2022). Technical report, Orbis Research, 01
- 13. Grand View Research (2016) Business Process Management (BPM) Market Analysis By Solution (Automation, Process Modelling, Content & Document Management, Monitoring & Optimization, Integration), By Application (BFSI, IT & Telecom, Retail, Manufacturing, Healthcare, Government & Defense), By Deployment (Cloud-based, On-Premise), By End-User (SME, Large Enterprise) And Segment Forecasts To 2024. Technical report, Grand View Research, 01
- Van Der Aalst WMP (2013) Business process management: a comprehensive survey. ISRN Software Engineering, 2013

- Object Management Group (2013) Business Process Model and Notation (BPMN), v2.0.2. Available at: http://www.omg.org/spec/BPMN/2.0.2/
- 16. Alfresco Software, Inc. Activiti BPM Software Home. Available at: https://www.activiti.org
- 17. Camunda services GmbH Camunda BPM Platform. Available at: https://camunda.org
- 18. Signavio GmbH Signavio Business Transformation Suite. Available at: www.signavio.com
- Weske M (2012) Business process management architectures. In: Business Process Management: Concepts, Languages, Architectures, pp 333–371. Springer, Berlin
- 20. Dumas M, La Rosa M, Mendling J, Reijers HA et al (2013) Fundamentals of business process management, volume 1. Springer
- Object Management Group (2016) Decision Model and Notation (DMN), v1.1. Available at: http:// www.omg.org/spec/DMN/1.1/
- 22. Hürsch WL, Lopes CV (1995) Separation of concerns. Technical report
- National Heart Lung and Blood Institute and others (2016) Global strategy for diagnosis management and prevention of COPD (updated 2016)
- 24. ATS statement (2002) Guidelines for the Six-Minute walk test. Am J Respir Crit Care Med 166(1):111–117
- 25. Object Management Group (2016) Case Management Modeling Notation (CMMN), v1.1. Available at: http://www.omg.org/spec/CMMN/1.1/
- Workflow Management Coalition (2012) XML Process Definition Language. Available at: http:// www.xpdl.org
- Mendling J, Reijers HA, van der Aalst WMP (2010) Seven process modeling guidelines (7PMG). Inf Softw Technol 52(2):127–136
- Kiepuszewski B, ter Hofstede AHM, Bussler CJ (2000) On structured workflow modelling. In: International Conference on Advanced Information Systems Engineering, pp 431–445. Springer
- Dumas M, García-Bañuelos L, Polyvyanyy A (2010) Unraveling unstructured process models. In: Business Process Modeling Notation, volume 67 of Lecture Notes in Business Information Processing, pp 1–7. Springer, Berlin
- Batoulis K, Meyer A, Bazhenova E, Decker G, Weske M (2015) Extracting decision logic from process models. In: Advanced Information Systems Engineering, pp 349–366. Springer
- 31. Debevoise T, Taylor J, Sinur J, Geneva R (2014) The MicroGuide to Process and Decision Modeling in BPMN/DMN: Building More Effective Processes by Integrating Process Modeling with Decision Modeling. CreateSpace Independent Publishing Platform
- 32. Halle BV, Goldberg L (2009) The decision model: a business logic framework linking business and technology. CRC Press
- Dingsøyr T, Nerur SP, Balijepally V, Moe NB (2012) A decade of agile methodologies: Towards explaining agile software development. J Syst Softw 85(6):1213–1221
- 34. Kettinger WJ, Teng James TC, Guha S (1997) Business process change: a study of methodologies, techniques, and tools. MIS quarterly 21(1):55–80
- 35. Khodambashi S (2013) Business process re-engineering application in healthcare in a relation to health information systems. Procedia Technology 9:949–957
- Ciccio CD, Marrella A, Russo A (2015) Knowledge-intensive processes: Characteristics, requirements and analysis of contemporary approaches. Journal on Data Semantics 4(1):29–57
- Van der Aalst WMP, ter Hofstede AHM, Kiepuszewski B, Barros AP (2003) Workflow patterns. Distributed and Parallel Databases 14(1):5–51
- Russell N, Ter Hofstede AHM, Edmond D, Van der Aalst WMP (2005) Workflow data patterns: Identification, representation and tool support. In: International Conference on Conceptual Modeling, pp 353–368. Springer
- Lanz A, Weber B, Reichert M (2010) Workflow time patterns for process-aware information systems. In: Enterprise, Business-Process and Information Systems Modeling, pp 94–107. Springer
- Rotar AM, Van den Berg MJ, Kringos DS, Klazinga NS (2016) Reporting and use of the OECD health care quality indicators at national and regional level in 15 countries. Int J Qual Health Care 28(3):398–404
- 41. Biard T, Le Mauff A, Bigand M, Bourey J-P (2015) Separation of decision modeling from business process modeling using new Decision Model and Notation (DMN) for automating operational decision-making. In: Risks and Resilience of Collaborative Networks, pp 489–496. Springer

- 42. Von Halle B, Goldberg L (2010) The decision model taylor and francis group
- 43. Shiffman RN, Greenes RA (1994) Improving clinical guidelines with logic and decision-table techniques: application to hepatitis immunization recommendations. Med Dec Making 14(3):245–254
- 44. Calvanese D, Dumas M, Laurson Ü, Maggi FM, Montali M, Teinemaa I (2016) Semantics and analysis of DMN decision tables. In: Business Process Management: 14th International Conference, BPM 2016, pp 217–233. Springer International Publishing, Rio de Janeiro
- 45. Zerbato F, Oliboni B, Combi C, Campos M, Juarez JM (2015) BPMN-Based Representation and Comparison of Clinical Pathways for Catheter-Related bloodstream infections. In: Balakrishnan P, Srivatsava J, Fu W-T, Harabagiu SM, Wang F (eds) 2015 International Conference on Healthcare Informatics, ICHI 2015. IEEE Computer Society, Dallas, pp 346–355
- 46. Rosselli Del Turco M, Ponti A, Bick U, Biganzoli L, Cserni G, Cutuli B, Decker T, Dietel M, Gentilini O, Kuehn T, Mano MP, Mantellini P, Marotti L, Poortmans P, Rank F, Roe H, Scaffidi E, van der Hage JA, Viale G, Wells C, Welnicka-Jaskiewicz M, Wengstöm Y, Cataliotti L (2010) Quality indicators in breast cancer care. Eur J Cancer 46(13):2344–2356
- 47. Peleg M, Tu S, Manindroo A, Altman RB (2004) Modeling and analyzing biomedical processes using workflow/Petri Net models and tools. In: MEDINFO 2004 - Proceedings of the 11th World Congress on Medical Informatics, volume 107 of Studies in Health Technology and Informatics, pp 74–78. IOS Press, San Francisco
- Lerner BS, Christov S, Wise A, Osterweil LJ (2008) Exception handling patterns for processes. In: Proceedings of the 4th international workshop on exception handling, pp 55–61. ACM
- 49. Kumar A, Barton RR (2017) Controlled violation of temporal process constraints models, algorithms and results. Inf Syst 64:410–424
- 50. Kunze M, Weske M (2016) Behavioural Models: From modelling finite automata to analysing business processes. Springer
- Combi C, Gozzi M, Juarez J, Oliboni B, Pozzi G (2007) Conceptual modeling of temporal clinical workflows. In: IEEE 14th International Symposium on Temporal Representation and Reasoning, pp 70–81. IEEE
- Lanz A, Weber B, Reichert M (2014) Time patterns for process-aware information systems. Requir. Eng. 19(2):113–141
- 53. Lanz A, Posenato R, Combi C, Reichert M (2016) Controlling time-awareness in modularized processes. In: Proceedings of the 17th international conference enterprise, business-process and information systems modeling BPMDS, Ljubljana, Slovenia, pp 157–172
- 54. Lanz A, Reichert M, Weber B (2016) Process time patterns: A formal foundation. Inf. Syst. 57:38-68
- Combi C, Oliboni B, Zerbato F (2017) Modeling and handling duration constraints in BPMN 2.0. In: Proceedings of the Symposium on Applied Computing, SAC '17, pp 727–734. ACM, Marrakech
- 56. Combi C, Sala P, Zerbato F (2017) Driving time-dependent paths in clinical BPMN processes. ACM, Marrakech
- 57. Bartolome R, Celli WATS, MacNee AATS, Agusti AA et al (2004) Standards for the diagnosis and treatment of patients with COPD: a summary of the ATS/ERS position paper. Eur Respir J 23(6):932–946
- Romain A, Pauwels A, Buist S, Calverley PMA, Jenkins CR, Hurd SS (2001) Global strategy for the diagnosis, management, and prevention of chronic obstructive pulmonary disease. Am J Respir Crit Care Med 163:1256–1276
- 59. Ortiz MLM, Morera J (2012) COPD: Differential Diagnosis INTECH open access publisher
- Price DB, Yawn BP, Jones RCM (2010) Improving the differential diagnosis of chronic obstructive pulmonary disease in primary care. Mayo Clin Proc 85(12):1122–1129
- Hardie JA, Sonia Buist A, Vollmer WM, Ellingsen I, Bakke PS, Mørkve O (2002) Risk of overdiagnosis of COPD in asymptomatic elderly never-smokers. Eur Respir J 20(5):1117–1122
- 62. personale SezioneControlligovernoeSSRSERSistemaEpidemiologicoRegionale (2016) Relazione Socio Sanitaria Della Regione del Veneto: Anno 2015. Technical report, Regione Veneto, 01
- Sezione Controlli governo e personale SSR SER Sistema Epidemiologico Regionale (2017) Relazione Socio Sanitaria Della Regione del Veneto: Anno 2016. Technical report, Regione Veneto, 01
- 64. Dal Negro R, Rossi A, Cerveri I The burden of copd in italy: results from the confronting copd survey. Respiratory Medicine, 97(Supplement C):S43 – S50, 2003. The economic impact of COPD in North America and Europe Analysis of the Confronting COPD survey

- Koleva D, Motterlini N, Banfi P, Garattini L (2007) Healthcare costs of COPD in Italian referral centres: A prospective study. Respir Med 101(11):2312–2320
- Guerriero M, Caminati M, Viegi G, Senna G, Cesana G, Pomari C (2015) COPD prevalence in a north-eastern italian general population. Respiratory medicine 109(8):1040–1047
- 67. Casas A, Troosters T, Garcia-Aymerich J, Roca J, Hernández C, Alonso A, del Pozo F, de Toledo P, Antó JM, Rodríguez-Roisín R et al (2006) Integrated care prevents hospitalisations for exacerbations in COPD patients. Eur Respir J 28(1):123–130
- 68. Combi C, Keravnou-Papailiou E, Shahar Y (2010) Temporal information systems in medicine, Springer Science & Business Media
- Norris SL, Glasgow RE, Engelgau MM, Os'Connor PJ, McCulloch. D (2003) Chronic disease management. Disease Management &, Health Outcomes 11(8):477–488
- Blackburn R, Stokes D (2000) Breaking down the barriers: using focus groups to research small and medium-sized enterprises. International Small Business Journal 19(1):44–67
- 71. Agency for Healthcare Research and Quality U.S. Department of Health and Human Services (2016) Prevention Quality Indicator 05 (PQI 05) Chronic Obstructive Pulmonary Disease (COPD) or Asthma in Older Adults Admission Rate, pp 1–6
- 72. National Committee for Quality Assurance (NCQA) Use of spirometry testing in the assessment and diagnosis of COPD: percentage of members 40 years of age and older with a new diagnosis of COPD or newly active COPD, who received appropriate spirometry testing to confirm the diagnosis, pp 1–2. 2015. Washington (DC)
- Camunda services GmbH Camunda DMN Simulator. Available at: https://camunda.org/dmn/ simulator/
- 74. Cohn D, Hull R (2009) Business artifacts: A data-centric approach to modeling business operations and processes. Bulletin of the IEEE Computer Society Technical Committee on Data Engineering 32(3):3–9
- 75. Dadam P, Reichert M (2009) The ADEPT project: a decade of research and development for robust and flexible process support. Computer Science-Research and Development 23(2):81–97
- Herzberg N, Kirchner K, Weske M (2014) Modeling and monitoring variability in hospital treatments: A scenario using CMMN. In: Business Process Management Workshops: International Workshops (BPM 2014), pp 3–15. Springer International Publishing, Eindhoven
- 77. Müller R, Rogge-Solti A (2011) BPMN for healthcare processes. In: Eichhorn D, Koschmider A, Zhang H (eds) Proceedings of the 3rd central-european workshop on services and their composition (ZEUS 2011), volume 705 of CEUR workshop proceedings, pp 65–72. CEUR-WS.org, Karlsruhe
- Braun R, Schlieter H, Burwitz M, Esswein W (2014) BPMN4CP: Design and implementation of a BPMN extension for clinical pathways. In: 2014 IEEE international conference on bioinformatics and biomedicine (BIBM), pp 9–16. IEEE
- Combi C, Oliboni B, Gabrieli A (2015) Conceptual modeling of clinical pathways making data and processes connected. In: Artificial intelligence in medicine - 15th conference on artificial intelligence in medicine (AIME 2015), pp 57–62. Springer International Publishing, Pavia
- Zerbato F, Oliboni B, Combi C, Campos M, Juarez JM (2015) BPMN-based representation and comparison of clinical pathways for catheter-related bloodstream infections. In: IEEE international conference on healthcare informatics (ICHI 2015), pp 346–355. IEEE
- Cánovas-Segura B, Campos M, Morales A, Juarez JM, Palacios F (2016) Development of a clinical decision support system for antibiotic management in a hospital environment. Progress in Artificial Intelligence 5(3):181–197
- Cánovas-Segura B, Zerbato F, Oliboni B, Combi C, Campos M, Morales Nicolás A, Juarez JM, Marín R, Palacios F (2017) A Process-Oriented Approach for Supporting Clinical Decisions for Infection Management. In: 2017 IEEE International Conference on Healthcare Informatics, ICHI 2017, Park City, UT, USA, August 23–26, 2017, pp 91–100
- Scheuerlein H, Rauchfuss F, Dittmar Y, Molle Rüdiger, Lehmann T, Pienkos N, Settmacher U (2012) New methods for clinical pathways—business process modeling notation (BPMN) and tangible business process modeling (t.BPM). Langenbeck's Arch Surg 397(5):755–761
- 84. Ferrante S, Bonacina S, Pozzi G, Pinciroli F, Marceglia S (2016) A design methodology for medical processes. Applied Clinical Informatics 7(1):191–210. https://doi.org/10.4338/ACI -2015-08-RA-0111. https://www.scopus.com/inward/record.uri?eid=2-s2.0-84962110212&partner ID=40&md5=74b633aceae934c56cd22dc0eaf2e409

- Bazhenova E, Weske M (2015) Deriving decision models from process models by enhanced decision mining. In: Business process management workshops, volume 256 of lecture notes in business information processing, pp 444–457. Springer
- Calvanese D, Dumas M, Laurson Ü, Maggi FM, Montali M, Teinemaa I (2016) Semantics and analysis of DMN decision tables. In: La Rosa M, Loos P, Pastor O (eds) 14th International conference on business process management (BPM 2016), pp 217–233, Cham. Springer International Publishing
- Bliemel M, Hassanein K (2004) E-health: applying business process reengineering principles to healthcare in Canada. Int J Electron Bus 2(6):625–643
- McNulty T (2002) Reengineering as knowledge management a case of change in UK healthcare. Manag Learn 33(4):439–458
- Peleg M (2013) Computer-interpretable clinical guidelines: a methodological review. J Biomed Inform 46(4):744–763
- 90. Panzarasa S, Quaglini S, Cavallini A, Marcheselli S, Stefanelli M, Micieli G (2007) Computerised guidelines implementation: Obtaining feedback for revision of guidelines, clinical data model and data flow. In: 11th conference on artificial intelligence in medicine, (AIME 2011), pp 461–466, Amsterdam. Springer, Berlin
- De Clercq PA, Blom JA, Korsten HHM, Hasman A (2004) Approaches for creating computerinterpretable guidelines that facilitate decision support. Artif Intell Med 31(1):1–27
- 92. Ciccarese P, Caffi E, Boiocchi L, Quaglini S, Stefanelli M (2004) A guideline management system. In: Fieschi M, Coiera EW, Li JY-C (eds) MEDINFO 2004 - proceedings of the 11th world congress on medical informatics, volume 107 of studies in health technology and informatics, pp 28–32. IOS Press, San Francisco
- 93. Peleg M, Boxwala AA, Ogunyemi O, Zeng Q, Tu S, Lacson R, Bernstam E, Ash N, Mork P, Ohno-Machado L et al (2000) GLIF3: the evolution of a guideline representation format, American Medical Informatics Association
- 94. Miksch S, Shahar Y, Johnson P (1997) Asbru: a task-specific, intention-based, and time-oriented language for representing skeletal plans. Open University, Milton Keynes
- Sutton DR, Fox J (2003) The syntax and semantics of the PROforma guideline modeling language. J Am Med Inform Assoc 10(5):433–443
- 96. Tu SW, Campbell JR, Glasgow J, Nyman MA, McClure R, McClay J, Parker C, Hrabak KM, Berg D, Weida T et al (2007) The SAGE guideline model: achievements and overview. J Am Med Inform Assoc 14(5):589–598
- 97. Shalom E, Shahar Y, Lunenfeld E (2016) An architecture for a continuous, user-driven, and datadriven application of clinical guidelines and its evaluation. Journal of Biomedical Informatics 59:130–148
- Peleg M, Shahar Y, Quaglini S (2014) Making healthcare more accessible, better, faster, and cheaper: the mobiguide project. European Journal of ePractice: Issue on Mobile eHealth 20:5–20
- Shvo AS, Peleg M, Parimbelli E, Quaglini S, Napolitano C (2016) Interplay between clinical guidelines and organizational workflow systems: experience from the mobiguide project. Methods Inf Med 55(6):488–494
- Dolin RH, Alschuler L, Boyer S, Beebe C, Behlen FM, Biron PV et al (2006) HL7 clinical document architecture, release 2. J Am Med Inform Assoc 13(1):30–39
- 101. (1998). World Health Organization and Practice Management Information Corporation. ICD-9-CM: International Classification of Diseases, 9th Revision: Clinical Modification, volume 1 PMIC (Practice Management Information Corporation)
- 102. Donnelly K (2006) SNOMED-CT: The advanced terminology and coding system for eHealth. Studies in health technology and informatics 121:279
- 103. Lee W-N, Tu SW, Das AK (2009) Extracting cancer quality indicators from electronic medical records: Evaluation of an ontology-based virtual medical record approach. In: AMIA 2009, american medical informatics association annual symposium, San Francisco, CA, USA, November 14–18, 2009. AMIA
- 104. van Der Aalst WMP, Ter Hofstede AHM, Kiepuszewski B, Barros AP (2003) Workflow patterns. Distributed and parallel databases 14(1):5–51
- 105. Calvanese D, Giacomo GD, Montali M (2013) Foundations of data-aware process analysis: A database theory perspective. In: Proceedings of the 32 ACM SIGMOD symposium on principles of database systems, pp 1–12. ACM

- 106. Brantly ML, Paul LD, Miller BH, Falk RT, Wu M, Crystal RG (1988) Clinical features and history of the destructive lung disease associated with alpha-1-antitrypsin deficiency of adults with pulmonary symptoms. American review of respiratory disease 138(2):327–336
- 107. Miravitlles M, Andreu I, Romero Y, Sitjar S, Altés A, Anton E (2012) Difficulties in differential diagnosis of COPD and asthma in primary care. Br J Gen Pract 62(595):e68–e75
- Centers for Disease Control and Prevention and others (2013) International classification of diseases, ninth revision clinical modification (ICD-9-CM)
- Fagerstrom K-O, Schneider NG (1989) Measuring nicotine dependence: a review of the fagerstrom tolerance questionnaire. J Behav Med 12(2):159–182
- 110. Rodriguez-Roisin R (2000) Toward a consensus definition for COPD exacerbations. Chest 117(5):398S-401S
- 111. Garcia-Aymerich J, Hernandez C, Alonso A, Casas A, Rodriguez-Roisin R, Anto JM, Roca J (2007) Effects of an integrated care intervention on risk factors of COPD readmission. Respir Med 101(7):1462–1469
- 112. Guerrero M, Crisafulli E, Liapikou A, Huerta A, Gabarrús A, Chetta A, Soler N, Torres A (2016) Readmission for acute exacerbation within 30 days of discharge is associated with a subsequent progressive increase in mortality risk in COPD patients: a long-term observational study. Plos one 11(3):e0150737